Southern California Edison

Application for Surrender of License Borel Hydroelectric Project FERC Project No. 382

Volume II



Decommissioning Plan



May 2023

PAGE INTENTIONALLY LEFT BLANK

Southern California Edison

Application for Surrender of License Borel Hydroelectric Project FERC Project No. 382

Volume II

Decommissioning Plan

Southern California Edison 1515 Walnut Grove Ave. Rosemead, CA 91770

May 2023

PAGE INTENTIONALLY LEFT BLANK

Table of Contents

1	Introduction	1
2	Decommissioning Plan	5
2.1	Borel Project Segmentation	5
2.2 2.2.1	Segment 1 – Diversion Dam and Intake Structure to Tilley No.1 Concrete Flume Existing Features	8 8
2.2.2	SCE Proposal	11
2.3 2.3.1	Segment 2 – Tilley No. 1 Concrete Flume to Tilley No. 3 Concrete Flume Existing Features	12 12
2.3.2	SCE Proposal	15
2.4 2.4.1	Segment 3 – Tilley No. 3 Concrete Flume to End of SCE Land Existing Features	16 16
2.4.2	SCE Proposal	19
2.5 2.5.1	Segment 4 – End of SCE Land to Auxiliary Dam (Corps) Existing Features	21 21
2.5.2	SCE Proposal	24
2.6 2.6.1	Segment 5 – Auxiliary Dam (Corps) Existing Features	25 25
2.6.2	SCE Proposal	26
2.7 2.7.1	Segment 6 – Auxiliary Dam to Alta Sierra Avenue Existing Features	26 26
2.7.2	SCE Proposal	32
2.8 2.8.1	Segment 7 – Alta Sierra Avenue to Erskine Steel Flume Existing Features	33 33
2.8.2	SCE Proposal	35
2.9 2.9.1	Segment 8 – Erskine Steel Flume to Bodfish Siphon Existing Features	36 36
2.9.2	SCE Proposal	38
2.10 2.10.1	Segment 9 – Bodfish Siphon to Pioneer Steel Siphon Existing Features	39 39
2.10.2	SCE Proposal	42
2.11 2.11.1	Segment 10 – Pioneer Steel Siphon to Forebay Structure Existing Features	43 43
2.11.2	SCE Proposal	47
2.12 2.12.1	Segment 11 – Forebay Structure to Tailrace Existing Features	48 48
2.12.2	SCE Proposal	53

3	Estimated Schedule	57
4	Proposed Measures	59
5	References	72

List of Figures

Figure 1-1. Borel Project and Land Ownership	3
Figure 2-1. Borel Project – Decommissioning Segments	7
Figure 2-2. Aerial View of the Diversion Dam and Intake Structure, Looking Downstream (Photo Date: 09/2012)	8
Figure 2-3. Canal Inlet Structure, Looking Downstream (Photo Date: 06/2013)	9
Figure 2-4. Canal Inlet Structures and Concrete-lined Canal, Looking Upstream. (Photo Date: 02/2021)	9
Figure 2-5. Canal from Canal Inlet Structure to Tilley No. 1 Concrete Flume (out of frame)(looking downstream). (Photo Date: 02/2021)	10
Figure 2-6. Storehouse and Maintenance Yard. (Photo Date: 03/2022)	10
Figure 2-7. Tilley No. 1 Concrete Flume, Looking Upstream (Photo Date: 02/2021)	13
Figure 2-8. Concrete-lined Canal and Bridge, Looking Downstream (Downstream of Tilley No. 1 Concrete Flume) (Photo Date: 02/2021)	13
Figure 2-9. Canal Filled with Sedimentation, Looking Upstream (Downstream of Tilley No. 2 Concrete Flume) (Photo Date: 03/2022)	14
Figure 2-10. Tilley No. 2 Concrete Flume, Looking Downstream (Photo Date: 02/2021)	14
Figure 2-11. Tilley No. 3 Concrete Flume, Looking Downstream (Photo Date: 02/2021)	15
Figure 2-12. Concrete-lined Canal Downstream of Tilley No. 3 Concrete Flume, Looking Downstream (Photo Date: 02/2021)	17
Figure 2-13. Typical Embankment with Rock Slope Protection, Looking Upstream (Photo Date: 02/2021)	17
Figure 2-14. Unnamed Steel Bridge near Station 89+00 (Photo Date: 02/2021)	18
Figure 2-15. Refugio Concrete Box Flume (near Station 149+00), Looking Upstream (Photo Date: 02/2021)	18
Figure 2-16. Bailey Bridge, Stored on Federal Lands Administered by the Forest Service (Photo Date: 02/2021)	19
Figure 2-17. Example of Recreational Roads on Federal Lands Administered by the Forest Service that may be used for Construction Access. (Photo Date: 03/2022)	21
Figure 2-18 Concrete-lined Canal Looking Downstream (Photo Date: 02/2021)	21
Figure 2-19, Rock Slope Protection on Canal Bank (Photo Date: 02/2021)	
Figure 2-20. Sawmill Bridge at Station 173+50, Looking Downstream (Photo Date: 02/2021)	
Figure 2-21. Rich Gulch Concrete Siphon Exit, Looking Upstream (Photo Date: 03/2022)	
Figure 2-22. Kern River Concrete Siphon Entrance, Looking Downstream (Photo Date: 03/2022)	
Figure 2-23. Engineers Point, Looking Upstream (Photo Date: 02/2021)	
Figure 2-24. Limits of Corps Canal Backfill and Auxiliary Dam, Looking Downstream (Photo Date: 02/2021)	

Figure 2-25.	Federal Lands Administered by the Corps with Filled Canal and Auxiliary Dam, Looking Upstream (Photo Date: 01/2020)	26
Figure 2-26.	Concrete-lined Canal, Looking Downstream (Photo Date: 02/2021)	27
Figure 2-27.	Eva Avenue (Photo Date: 03/2023)	28
Figure 2-28.	SR 178 Bridges (Photo Date: 03/2022)	28
Figure 2-29.	Lake Isabella Boulevard Bridge, Looking Upstream (Photo Date: 03/2022)	29
Figure 2-30.	Lakeland Walk Bridge (Non-Project), Looking Downstream (Photo Date: 03/2022)	29
Figure 2-31.	School House No. 1 Concrete Flume, Looking Downstream (Photo Date: 02/2021)	30
Figure 2-32.	School House Cut, Looking Downstream (Photo Date: 03/2022)	30
Figure 2-33.	School House Cut Flume, Looking Upstream (Photo Date: 02/2021)	31
Figure 2-34.	Erskine Creek Road Bridge, Looking Downstream (Photo Date: 02/2021)	31
Figure 2-35.	Concrete-lined Canal, Looking Downstream (Photo Date: 02/2021)	33
Figure 2-36.	School Pedestrian Bridge, Looking Downstream (Photo Date: 01/2020)	34
Figure 2-37.	County Emergency Vehicular Bridges (Non-Project) (Photo Date: 01/2020)	34
Figure 2-38.	Erskine Creek Flume, Looking Upstream (Photo Date: 01/2020)	35
Figure 2-39.	Concrete-lined Canal, Looking Downstream (Photo Date: 01/2020)	37
Figure 2-40.	Bodfish Siphon, Looking Downstream (Photo Date: 01/2020)	37
Figure 2-41.	Location of Proposed Biofiltration Basin at Bodfish Siphon (Photo Date: 03/2022)	39
Figure 2-42.	Concrete-lined Canal, Looking Upstream (Photo Date: 03/2022)	40
Figure 2-43.	Retaining Wall Structure Downstream of Tunnel No. 1, Looking Downstream (Photo Date: 01/2020)	40
Figure 2-44.	Tunnel No. 1 Entrance, Looking Downstream (Photo Date: 06/2020)	41
Figure 2-45.	Pioneer Steel Siphon, Looking Upstream (Photo Date: 02/2021)	41
Figure 2-46.	Pioneer Steel Siphon Dewatering and Drainage Features (Photo Date: 02/2021)	42
Figure 2-47.	Concrete-lined Canal, Looking Downstream of Tunnel No. 1 1/2 (Photo Date: 02/2021)	44
Figure 2-48.	Tunnel No. 1 1/2 Entrance, Looking Downstream (Photo Date: 02/2021)	44
Figure 2-49.	Tunnel No. 2 Entrance, Looking Downstream (Photo Date: 02/2021)	45
Figure 2-50.	Utility Poles on Right Bank Downstream of Tunnel No. 2, Looking Downstream (Photo Date: 02/2021)	45
Figure 2-51.	Flume No. 623, Looking Downstream (Photo Date: 02/2021)	46
Figure 2-52.	Tunnel No. 3 Exit, Looking Upstream (Photo Date: 02/2021)	46
Figure 2-53.	Profanity Steel Flume, Looking Downstream (Photo Date: 02/2021)	47
Figure 2-54.	Forebay Structure (Photo Date: 02/2021)	49
Figure 2-55.	Penstocks (Photo Date: 03/2022)	49
Figure 2-56.	Powerhouse Interior (Photo Date: 01/2020)	50
Figure 2-57.	Storage Building (Photo Date: 03/2022)	51
Figure 2-58.	Pump House (Photo Date: 03/2022)	51
Figure 2-59.	Restroom/Outbuilding at Powerhouse (Photo Date: 03/2022)	52
Figure 2-60.	Borel Switchyard and Powerhouse (Photo Date: 03/2022)	52
Figure 2-61.	Tailrace Outlet at Kern River (Photo Date: 03/2022)	53
Figure 2-62.	Existing Crane (Photo Date: 01/2020)	55

Figure 2-63.	Powerhouse Access Road (Photo Date: 03/2022)	56
Figure 2-64.	Staging Area for Penstock Demolition (Mid-slope) (Photo Date: 03/2022)	56

List of Tables

Table 2-1. Borel Project Decommissioning Segments	6
Table 4-1. Measures for Borel Project Decommissioning	60

Appendices

Appendix A: Stormwater and Drainage Report

Appendix B: Borel Project Segment Map Set

Appendix C: Decommissioning Plan 30% Plans

Appendix D: Estimated Schedule

Acronyms and Abbreviations

°C	Degrees Celsius
°F	Degrees Fahrenheit
1D	One-dimensional
2D	Two-dimensional
ac-ft	Acre-feet
ACHP	Advisory Council on Historic Preservation
AIRs	Additional Information Requests
AOU	American Ornithologists Union
APDEA	Applicant-Prepared Draft Environmental Assessment
APE	Area of Potential effect
Auxiliary Dam	Lake Isabella Auxiliary Dam, U.S. Army Corps of Engineers
BA	Biological Assessment
BGEPA	Bald and Golden Eagle Protection Act
BLM	U.S. Department of the Interior, Bureau of Land Management
BLM-S	BLM Sensitive Species
BMP	Best Management Practice
BO2	Blue Oak Savanna
Borel Project	Borel Hydroelectric Project, FERC Number 382
C.F.R.	Code of Federal Regulations
cal YBP	Calendar Years Before Present
CCMP	California Coastal Management Program
CDFA	California Department of Food and Agriculture
CDFG	California Department of Fish and Game
CDFW	California Department of Fish and Wildlife
CEII	Critical Energy Infrastructure Information
CEQA	California Environmental Quality Act

CESA	California Endangered Species Act
CFGC	California Fish and Game Commission
CFR	Code of Federal Regulations
Cfs	Cubic Feet per Second
CGS	California Geological Survey
CNDDB	California Natural Diversity Data Base
CNPS	California Native Plant Society
Corps	U.S. Army Corps of Engineers
CRPR	California Rare Plant Rank
CRWQCB	California Regional Water Quality Control Board
CSB	California Stream Bioassessment
CUI	Controlled Unclassified Information
CWA	Clean Water Act
CWHR	California Wildlife Habitat Relationships
CZMA	Coastal Zone Management Act
DBH	Diameter at Breast Height
DEM	Digital Elevation Model
DO	Dissolved Oxygen
DPS	Distinct Population Segment
EA	Environmental Assessment
EAP	Emergency Action Plan
e-cat	Cataraft Electrofisher
EPT	Ephemeroptera (mayflies), Plecoptera (stoneflies), and Trichoptera (caddisflies)
ESA	Endangered Species Act
ESRI	Environmental Systems Research Institute
FAQs	Frequently Asked Questions
FERC	Federal Energy Regulatory Commission (or Commission)

FGV	Fine-grained Volcanic
Forest Service	U.S. Department of Agriculture, Forest Service
FP	Fully Protected Species
FPA	Federal Power Act
FSS	Forest Service Sensitive Species
ft	Feet
FW	Forest Service Watch List Species
FYLF	Foothill Yellow-legged Frog
GPS	Global Positioning System
H:V	Horizontal to Vertical
HABS	Historic American Buildings Survey
HAER	Historic American Engineering Record
HEC-HMS	Hydrologic Engineering Center's Hydrologic Modeling System
HEC-RAS	Hydrologic Engineering Center Riverine Analysis System
HHRD	Historic Human Remains Detection
HP	Horsepower
Hz	Hertz
ICF	Institute for Canine Forensics
IPaC	Information for Planning and Consultation
IRWMP	Integrated Regional Water Management Plan
KCF	Kern Canyon Fault
KOP	Key Observation Point
KR&LAEP	Kern River and Los Angeles Electric Power Company
KR-1	Southern California Edison's Kern River No. 1 Project
KRVHS	Kern River Valley Historical Society
KRVM	Kern River Valley Museum
Ksat	Saturated Hydraulic Conductivity of Soil

kV	Kilovolt
kW	Kilowatt
Lidar	Light Detection and Ranging
Lower Borel	Downstream of the Auxiliary Dam
LRMP	Land and Resource Management Plan
msl	Mean Sea Level
MW	Megawatt
NAD83	North American Datum of 1983
NAHC	Native American Heritage Commission
NAVD 88	North American Vertical Datum of 1988
NED	National Elevation Dataset
NEPA	National Environmental Policy Act
NFS	National Forest System
NHPA	National Historic Preservation Act
NLCD	National Land Cover Database
NNIP	Non-native Invasive Plant
NOAA	National Oceanic and Atmospheric Administration
NRCS	Natural Resources Conservation Service
NRHP	National Register of Historic Places
OHV	Off Highway Vehicle
O&M	Operations and Maintenance
РСТ	Pacific Crest Trail
PL&P	Pacific Light and Power Company
Plan	Decommissioning Plan
Plan Set	30% Design Plans
Powerhouse	Borel Powerhouse
PT&L	Power, Transit, and Light Company

QSD	Qualified SWPPP Developer
QSP	Qualified SWPPP Practitioner
RM	River Mile
RMP	Resource Management Plan
RMZ	Recreational Management Zones
ROS	Recreation Opportunity Spectrum
RV	Recreational Vehicle
SCE	Southern California Edison Company
SCS	Soil Conservation Service
SHPO	State Historic Preservation Office(r)
SIO	Scenic Integrity Objective
SMA	Special Management Area
SMS	Scenery Management System
SQF	Sequoia National Forest
SR	State Route
SRMA	Special Recreation Management Area
SSC	Species of Special Concern
SSJVIC	Southern San Joaquin Valley Information Center
SSURGO	Soil Survey Geographic Database
Surrender Application	Application for Surrender of License
Switchyard	non-Project Borel Switchyard
SWPPP	Stormwater Pollution Prevention Plan
SWRCB	State Water Resources Control Board
TCE	Temporary Construction Easement
TCL	Traditional Cultural Landscape
ТСР	Traditional Cultural Property
TIN	Triangular Irregular Network

U.S.	United States
U.S.C.	United States Code
Upper Borel	Upstream of the Auxiliary Dam
USDA	U.S. Department of Agriculture
USEPA	U.S. Environmental Protection Agency
USFWS	U.S. Fish and Wildlife Service
USGS	U.S. Geological Survey
V	Volt
VegCAMP	Vegetation Classification and Mapping Program
VMS	Visual Management System
VQO	Visual Quality Objective
VRM	Visual Resource Management
WEAP	Worker Environmental Awareness Program
YOY	Young of the Year

1 Introduction

SCE has prepared this Decommissioning Plan (Plan) as part of its Application to FERC for surrender of the Borel Project license. This Plan is prepared in conformance with Title 18 C.F.R.), Chapter I, Subchapter B – Regulations under the FPA, Part 6 – Surrender or Termination of License.

The Borel Project, which has a FERC-authorized installed capacity of 12,000 kilowatts (kW), is located within Kern County, California, and consists of: 1) a 158-foot-long, 4-foot-high concrete Diversion Dam on the North Fork of the Kern River; 2) a 61-foot-long Intake Structure with three 10-by-10-foot radial gates; 3) Canal Inlet Structures consisting of a canal intake, trash racks, and a sluice gate; 4) Borel Canal, consisting of 1,985 feet of tunnel, 1,651 feet of steel-lined flume, 51,835 feet of concrete-lined canal, and 3,683 feet of siphon at four locations (i.e., Rich Gulch, Kern River, Bodfish and Pioneer); 5) an auxiliary intake structure to the Borel Canal located at the Corps' Lake Isabella Auxiliary Dam; 6) a small off-channel, concrete lined forebay; and 7) four steel penstocks. Penstocks 1 and 2 are 526 feet long and 565 feet long respectively, with varying diameters between 42 and 60 inches, and Penstocks 3 and 4 each are 60 inches in diameter and extend 622 feet at which point they join together to form a single 84-inch-diameter, 94-foot-long penstock; 8) the Borel Powerhouse (Powerhouse) with two 3,000- kW generators and a 6,000-kW generator for a total installed capacity of 12,000 kW, and which discharge to the Kern River; and 9) other appurtenant facilities.

The Borel Project occupies approximately 363 acres of land, of which 199.41 acres are Federally owned, as follows: 159.24 acres are managed by the United States Department of Agriculture (USDA), Forest Service (Forest Service) as part of the Sequoia National Forest (SNF), 29.47 acres are administered by the U.S. Department of the Interior, Bureau of Land Management (BLM) and 10.7 acres are administered by the Corps (Figure 1-1). Private lands include those owned by SCE and other private party parcels. The Federal Power Commission, predecessor agency to FERC, issued an original license for the Borel Project in 1925. On May 17, 2006, FERC issued a new license to SCE, with a license term ending on April 30, 2046.

Figure 1-1 shows the Borel Project features and Borel FERC Project boundary, which are described in more detail in Section 2 and Appendix B, Borel Project Segment Map Set of this Plan as well as in Exhibit A, Project Description; Exhibit F, Design Drawings; and Exhibit G, Project Maps that are available in FERC's eLibrary and hereby made part of SCE's Surrender Application by reference.

Specifically, as originally licensed, the Borel Project used a diversion dam and intake structure on the North Fork of the Kern River to divert water into the Borel Canal for power generation at the Powerhouse. The Flood Control Act of 1944 authorized the Corps' Isabella project to construct the Isabella Dam and reservoir on the Kern River, mainly to provide flood protection for Bakersfield and the surrounding communities. The Corps began construction of the Isabella Dam in 1948, which was completed in 1953. The auxiliary dam was constructed over a portion of the Borel canal, and a concrete conduit was built through the dam to allow continued flow through the canal. The elements of the Borel system beneath and to the north of Isabella Dam were heavily modified during the dam's construction. When full, Lake Isabella impounds more than 110,000 acre-feet, and the diversion dam, overflow weir, intake structure, and 4.2 miles of the upper portion of the Borel Canal are inundated. In 2017, the Corps implemented a safety modification project to its Lake Isabella Auxiliary Dam for which the Corps condemned 10.7 acres of private and public land associated with the Borel

Project and sealed off the existing section of conduit through the Auxiliary Dam by filling it with concrete and abandoning the conduit in place. This action rendered the Borel Project nonfunctional, requiring SCE to file an application to surrender the Borel Project license. Consequently, the Commission, by letter dated March 11, 2019, directed SCE to prepare an application to surrender its license for the Borel Project, in accordance with Section 6.1 of its regulations. SCE holds water right license No. 005731 (application No. 013778) to divert 605 cfs of water and use it for power. SCE intends to surrender its power water right at the appropriate time.

As part of the Surrender Application, this Plan consists of this introduction, Section 1, and the sections that follow:

- Section 2 describes the SCE-proposed plan to decommission the licensed Borel Project features. For ease of presentation, the Borel Project is divided into 11 canal segments; as such, Section 2 is divided into 11 corresponding subsections.
- Section 3 provides an estimated schedule for decommissioning and construction activities to take place once FERC issues its order approving the license surrender.
- Section 4 presents proposed measures that will follow the decommissioning considerations.
- Section 5 provides a list of citations for material/information referenced in the Plan.

Refer to Volume III, Applicant-Prepared Draft Environmental Assessment (APDEA), for an assessment of any effects of decommissioning on environmental, recreational, cultural, and other resources, and to Volume IV, Record of Consultation, for a description of consultation with stakeholders regarding the decommissioning of Borel Project facilities and surrender of the license. Volume V includes Privileged Information, including compiled data with regard to pre-contact and historic-era archaeological resources, historic built environment resources, Traditional Cultural Properties, and a record of consultation with Native American Tribes, as well as agencies as appropriate.



Figure 1-1. Borel Project and Land Ownership

PAGE INTENTIONALLY LEFT BLANK

2 Decommissioning Plan

The following subsections describe the proposed decommissioning actions for the Borel Project by segment. For spatial orientation, photos of the Borel Project presented in Section 2 include a blue arrow (bottom left corner), indicating the direction of water flow when the Borel Project was operating, if applicable, and a north arrow (bottom right corner).

2.1 Borel Project Segmentation

For purposes of this Plan and related analyses, the Borel Project was divided into 11 segments based on several conditions including:

- land ownership;
- construction action;
- access;
- location; and
- other common conditions.

A list of Borel Project segments is provided in Table 2-1.

Figure 2-1 shows the 11 Borel Project segments.

Segments 1 through 4 are located upstream from the Auxiliary Dam (Upper Borel) and entirely within the limits of Lake Isabella and subject to inundation when the water surface elevation of the reservoir is at an approximate elevation of 2,550 ft or higher. Access to these segments and decommissioning actions will be dependent upon water year and lake levels. **Segment 5** is located partially within the reservoir and partially downstream of the Auxiliary Dam (Lower Borel). **Segments 6 through 11** are located downstream from the Auxiliary Dam (Lower Borel), and access is not impacted by reservoir operations.

A series of Borel Project Segment Maps are provided in Appendix B for reference. Appendix C includes a 30% Design Plan Set (Plan Set) organized into the following sections:

- G-Series: Cover sheet, Sheet Index, and Borel Project Information
- V-Series: Aerial Photos, Property Ownership, and Site Access
- C100 Series: Civil Improvements (Plan view and sections)
- C300 Series: Typical Sections
- C500 Series: Details

Throughout the following descriptions of proposed actions for Borel Project segments, reference is made to fill materials for purposes of backfilling and grading. Fill materials may be sourced from onsite or offsite. Onsite fill material is native material and will be used provided the material is free of materials larger than six inches in dimension, organic material, and other deleterious materials. Offsite fill material will be chemically screened prior to use and be weed free. Fill material is discussed in Section 4 – Proposed Measures of this Plan and will also be addressed in the Borel Project construction specifications.

Geotechnical investigations and earthworks specifications will be developed at subsequent phases of design that will define material properties for engineered fill, maximum lift thickness, optimum moisture content, and compaction requirements.

Name	Description	Station ^a Number – Start	Station Number – End	Length (feet)
Segment 1	Diversion Dam and Intake Structure to Tilley No. 1 Flume	32+30	52+00	1,970
Segment 2	Tilley No. 1 Concrete Flume to Tilley No. 3 Concrete Flume	52+00	88+15	3,615
Segment 3	Tilley No. 3 Concrete Flume to end of SCE Land	88+15	161+00	7,285
Segment 4	End of SCE Land to Auxiliary Dam	161+00	273+80	11,280
Segment 5	Auxiliary Dam	273+80	298+00	2,420
Segment 6	Auxiliary Dam to Alta Sierra Avenue	298+00	398+48	10,048
Segment 7	Alta Sierra Avenue to Erskine Steel Flume	398+48	451+20	5,272
Segment 8	Erskine Steel Flume to Bodfish Siphon	451+20	506+29	5,509
Segment 9	Bodfish Siphon to Pioneer Steel Siphon	506+29	548+79	4,250
Segment 10	Pioneer Steel Siphon to Forebay Structure	548+79	628+74	7,995
Segment 11	Forebay Structure to Tailrace ^b	628+74	639+03	1,029
	Total			60,673

Table 2-1. B	orel Proiect	Decommissi	onina Se	aments
	01011101000	Decominison		ginento

^a Stationing is a form of measurement used by engineers and surveyors to show the linear distance from the point of origin. Each station is equal to one hundred feet from the linear point of origin. For example, xx + xx, where xx is multiplied by one hundred + xx is the additional number of feet. "32 + 30" is 3200+30 = 3,230 feet from the linear point of origin.

^b End stationing for this segment at terminus of spillway channel. Borel Project stationing was not provided in the 2006 License Exhibits (F, G, K) for the Penstocks and Powerhouse.



Figure 2-1. Borel Project – Decommissioning Segments

2.2 Segment 1 – Diversion Dam and Intake Structure to Tilley No.1 Concrete Flume

2.2.1 Existing Features

Segment 1 extends from the Diversion Dam and Intake Structure (Station 32+30) to Tilley No. 1 Concrete Flume (Station 52+00), approximately 1,970 feet. The Diversion Dam and Intake Structure and earthen channel from the Diversion Dam and Intake Structure to the canal inlet are on property owned by SCE. The canal inlet, canal, and Storehouse are on Federal lands administered by the Forest Service. The following subsections describe Borel Project features within this segment.

2.2.1.1 Diversion Dam and Intake Structure

The Diversion Dam and Intake Structure are located within Kern County on the North Fork Kern River. The Diversion Dam is a 4-foot-high, 158-foot-long mass concrete structure. The Intake Structure is approximately 61 feet long, with three 10-foot by 10-foot radial gates. An aerial view of the Diversion Dam and Intake Structure captured in 2012 is shown in Figure 2-2, with the Intake Structure visible and the Diversion Dam submerged underwater.



Figure 2-2. Aerial View of the Diversion Dam and Intake Structure, Looking Downstream (Photo Date: 09/2012)

An approximately 3,200-foot-long earthen berm extends from the Diversion Dam to the Canal Inlet Structure. A settling pond with an overflow dam that has a concrete section with two 8-foot by 10-foot radial gates is located on the Overflow Dam at Settling Basin approximately 1,500 feet downstream of the Diversion Dam and Intake Structure. The Diversion Dam and Intake Structure and channel are almost completely covered with sediment and overgrown with vegetation.

2.2.1.2 Canal Inlet Structures

The Canal Inlet Structures consist of a screen structure, sluice gate structure, and canal inlet structures. The screen structure is approximately 60 feet wide, with four openings covered with bar screens that allow for water to enter the canal from the Diversion Dam and Intake Structure channel. The sluice gate structure extends east from the end of the screen structure, consisting of four 9-footwide bays with sluice gates and stop logs. A concrete wing wall extends farther to the east. The screen and sluice gate structures are surrounded by dense vegetation, as shown in Figure 2-3.



Figure 2-3. Canal Inlet Structure, Looking Downstream (Photo Date: 06/2013)

The Canal Inlet Structures are located at the head of the canal and are approximately 40 feet wide and 30 feet long. An approximately 40-foot by 14-foot concrete deck is on top of the structure. Three manually operated steel radial gates are within the structure. Chain-link fencing lines the top of the structures on all sides to prevent access (Figure 2-4).



Figure 2-4. Canal Inlet Structures and Concrete-lined Canal, Looking Upstream. (Photo Date: 02/2021)

2.2.1.3 Concrete-lined Canal

A concrete-lined canal extends between the Canal Inlet Structure and Tilley No. 1 Concrete Flume, a distance of approximately 1,915 feet. The canal is a trapezoidal, concrete-lined structure with side slopes between 1.7 and 1.9H:1V and an average bottom width of approximately 22 feet. The amount of sedimentation and vegetation in the bottom of the canal varies greatly across the segment, (Figure 2-5). Based on available as-built drawings, the concrete liner is approximately 9 inches thick, with welded wire fabric reinforcement.



Figure 2-5. Canal from Canal Inlet Structure to Tilley No. 1 Concrete Flume (out of frame)(looking downstream). (Photo Date: 02/2021)

2.2.1.4 Storehouse

A SCE-owned storage building named the Storehouse is located adjacent to the Intake Structure. The Storehouse structure and adjacent maintenance yard are surrounded by chain-link fence (Figure 2-6). The adjacent utility line is not within the enclosed chain-link fence line.



Figure 2-6. Storehouse and Maintenance Yard. (Photo Date: 03/2022)

2.2.1.5 Access Roads

A public access road leads to the Intake Structure from East Evans Road, which is accessible from Wofford Boulevard/State Route (SR) 155. The paved upper portion of East Evans Road extends through a residential area and is under Kern County jurisdiction. The lower portion of East Evans Road that leads to the Storehouse, Canal Inlet Structure, and canal is on Federal lands administered by the Forest Service.

2.2.2 SCE Proposal

The proposed decommissioning actions and the relevant considerations for **Segment 1** are summarized below. Preliminary design plans reflecting the proposed actions are included in Appendix C, Sheet C-101.

2.2.2.1 Decommissioning Approach

The Plan consists of the following actions:

- **Diversion Dam and Intake Structure Abandon in Place**: No work is proposed at the Diversion Dam and Intake Structure as natural processes have partially buried the features and vegetation has grown around the structure. The canal intake screen and sluice gate structures will likewise be abandoned in place to avoid impacts on established vegetation and habitat.
- **Earthen Channel Abandon in Place**: No work is proposed along the former earthen channel leading from the Diversion Dam and Intake Structure to the Canal Inlet Structure. It is generally undetectable at this time and overgrown with vegetation.
- Overflow Dam at Settling Basin Abandon in Place: No work is proposed at the location of the Overflow Dam at Settling Basin. The concrete and steel structure is generally undetectable at this time and has become overgrown with vegetation.
- Canal Inlet Structure Abandon with Modification: The Canal Inlet Structure will be abandoned in place with modifications. Steel, trash racks, radial gates, slide gates, actuators, and fencing will be removed from the Canal Inlet Structure, and the vertical walls of the structure will be backfilled at a maximum slope of 3H:1V to daylight to adjacent ground. All waste will be hauled to an approved waste or recycling facility. If needed, the interior of the Canal Inlet Structure will be dewatered prior to fill placement.
- **Concrete-lined Canal Abandon with Modification**: The existing concrete canal will be backfilled with imported soil. The left bank¹ of the canal will remain in place and portions may be used for access. Ground disturbance will be kept to a minimum. The right bank of the canal will remain in place to avoid ground disturbance along the hillside. If needed, the interior of the canal will be dewatered prior to fill placement.

¹ For the purposes of this Plan, the terms "left bank" and "right bank" of the Borel Canal are used when looking downstream.

• Storehouse – Demolish and Haul Off Site: The existing Storehouse, materials stored inside the maintenance area, chain-link fence, and gates will be demolished and hauled off site. The foundation of the Storehouse structure will be left in place, and the surrounding area will be minimally graded to naturally conform to adjacent topography. The utility service to the Storehouse will be removed, but utility poles and overhead lines in the area will be protected in place to maintain service to adjacent parcels.

2.2.2.2 Access and Staging

Access to the Canal Inlet Structure is available via East Evans Road. The unpaved portion of East Evans Road to the terminus end at the Borel FERC Project boundary (approximately 800 linear feet) will be used for construction access and may therefore require minor improvements and grading during construction to accommodate construction vehicles. If improved, it is expected that the road will be abandoned in the improved condition. It is anticipated that all access roads for **Segment 1** will be within the limits of the concrete channel or along the left bank of the canal and access roads, which are within the Borel FERC Project boundary.

The area within the limits of the Storehouse fencing will be available for staging during construction and is approximately 0.3 acres. A second staging area will be located at the intersection of the East Evans Road terminus and the canal. The second staging area will be approximately 1.1 acres and require a temporary construction easement (TCE) as a portion of this staging area is outside the Borel FERC Project boundary and on Federal lands administered by the Forest Service. Staging areas developed for the sole purpose of construction activities will be graded back to preconstruction conditions upon completion of work.

2.3 Segment 2 – Tilley No. 1 Concrete Flume to Tilley No. 3 Concrete Flume

2.3.1 Existing Features

Segment 2 extends from the Tilley No. 1 Concrete Flume (Station 52+00) through the Tilley No. 3 Concrete Flume (Station 88+15), a distance of approximately 3,615 feet. All features within **Segment 2** are on property owned by SCE. The following subsections describe the Borel Project features within this segment.

2.3.1.1 Tilley No. 1 Concrete Flume

The Tilley No. 1 Concrete Flume (Figure 2-7) is a raised concrete flume approximately 115 feet long, including transitions. The rectangular flume is approximately 63 feet long, 16 feet wide, and 9 feet, 4 inches high. Design freeboard in the flume is approximately 1 foot, 5 inches. Concrete wingwalls extend on either side and transition to the trapezoidal lined canal.



Figure 2-7. Tilley No. 1 Concrete Flume, Looking Upstream (Photo Date: 02/2021)

2.3.1.2 Concrete-lined Canal

Segment 2 includes approximately 3,040 feet of lined canal. The canal is a trapezoidal, concretelined canal with side slopes averaging 1.8H:1V, with an average bottom width of approximately 22 feet. A 25-foot-wide bridge crosses the canal just downstream of the Tilley No. 1 Concrete Flume (Figure 2-8). The condition of the concrete liner varies greatly across the segment due to erosion and sedimentation (Figure 2-9). Based on available as-built drawings, the concrete liner is approximately 9 inches thick, with welded wire fabric reinforcement.



Figure 2-8. Concrete-lined Canal and Bridge, Looking Downstream (Downstream of Tilley No. 1 Concrete Flume) (Photo Date: 02/2021)

Portions of the canal within **Segment 2** have collapsed and/or are filled with sediment, as shown in Figure 2-9.



Figure 2-9. Canal Filled with Sedimentation, Looking Upstream (Downstream of Tilley No. 2 Concrete Flume) (Photo Date: 03/2022)

2.3.1.3 Tilley No. 2 Concrete Flume

The Tilley No. 2 Concrete Flume (Figure 2-10) is a raised concrete flume approximately 230 feet long, including transitions. The rectangular flume is approximately 153 feet, 6 inches long, 17 feet, 10 inches wide, and 11 feet high. Design freeboard in the flume is approximately 1 foot, 5 inches. Nine-inch channel beams cross the top of the flume approximately every 15 to 16 feet. Concrete transitions extend on either side, connecting to the trapezoidal lined canal.



Figure 2-10. Tilley No. 2 Concrete Flume, Looking Downstream (Photo Date: 02/2021)

2.3.1.4 Tilley No. 3 Concrete Flume

The Tilley No. 3 Concrete Flume (Figure 2-11) is a raised concrete flume approximately 230 feet long, including transitions (approx. 50 ft in length). The rectangular flume is 133 feet, 6 inches long, 18 feet wide, and 9 feet, 9 inches high. Design freeboard in the flume is approximately 1 foot, 3 inches. Nine-inch channel beams cross the top of the flume approximately every 15 to 16 feet. Concrete transitions extend on either side, connecting to the trapezoidal lined canal.



Figure 2-11. Tilley No. 3 Concrete Flume, Looking Downstream (Photo Date: 02/2021)

2.3.1.5 Access Roads

Features within **Segment 2** of the canal can be accessed on the road parallel to the canal in **Segment 1** and from a paved public road that extends approximately 1,200 feet from SR 155 adjacent to Tillie Creek Campground (non-Project recreation facility), which is on Federal lands administered by the Forest Service. From the edge of this paved road, the road splits into two unpaved roads and enters the lakebed: the first bends and extends to Station 79+00, downstream of the Tilley No. 2 Concrete Flume, and the second connects to the canal near Station 89+00, downstream of the Tilley No. 3 Concrete Flume, in **Segment 3**. The access roads in this area are seasonal, unmaintained, and only accessible when the reservoir water elevations permit. On a year-to-year basis they do not necessarily align with the Borel FERC Project boundary. Existing public unpaved dirt access roads run parallel to the canal extending from Tilley No. 1 Concrete Flume to the Tilley No. 2 Concrete Flume.

2.3.2 SCE Proposal

The proposed actions for **Segment 2** minimize ground disturbance and truck trips required and are summarized below. Preliminary design plans reflecting the proposed action are included in Appendix C, Sheets C-101 through C-103.

2.3.2.1 Decommissioning Approach

The Plan consists of the following actions:

- Tilley No. 1 Concrete Flume, Tilley No. 2 Concrete Flume, and Tilley No. 3 Concrete Flume – Abandon with Modification: Imported soil will be placed in the flume to the height of the wingwalls and will conform with the surface of the fill in the upstream and downstream canal sections. The outside walls of the flume on the waterside and landside will be backfilled and sloped at a maximum slope of 3H:1V to daylight into to the adjacent topography, appearing as an earthen embankment. Cofferdams and dewatering may be required to allow for construction access and backfill activities.
- **Concrete-lined Canal Abandon with Modification**: The concrete canal between STA 52+00 and 74+00 will be backfilled with imported soil. The banks of the canal will remain in

place and portions may be used for access. The concrete bridge deck at STA 55+50 will be removed and processed and blended with imported soil. The bridge foundations will remain in place. Ground disturbance will be kept to a minimum. If needed, the interior of the canal will be dewatered prior to fill placement.

• **Concrete-lined Canal – Abandon in Place**: No action will be required between STA 74+00 to 86+00 because this section of the canal has naturally filled with sediment (Figure 2-9).

2.3.2.2 Access and Staging

It is anticipated that access to **Segment 2** will use existing public access roads. However, the area on the north side of the canal is generally flat and subject to frequent recreational/vehicular traffic, and it is anticipated that the alignment of roadways will change through time depending on inundation and/or erosion. To minimize ground disturbance, SCE will work with the Forest Service to access the Borel Project area generally following the existing public access roads, but will likely use the roadways that are most prominent at the time of construction. These unpaved access roads will require temporary grading and other improvements for use by construction vehicles during construction. If improved, it is expected that they will be abandoned in the improved condition.

A potential staging area and additional access has been identified between the bend of the first access road to Station 79+00 and the canal within the Borel FERC Project boundary.

2.4 Segment 3 – Tilley No. 3 Concrete Flume to End of SCE Land

2.4.1 Existing Features

Segment 3 extends from the Tilley No. 3 Concrete Flume (Station 88+15) through the Refugio Concrete Box Flume to the end of SCE land (Station 161+100) along the Borel Project alignment, a distance of approximately 7,285 feet. All system features within **Segment 3** are on property owned by SCE. Access roads are located on Federal lands administered by the Forest Service. The following sections describe the Borel Project features within this segment.

2.4.1.1 Concrete-lined Canal

Segment 3 includes approximately 7,010 feet of concrete-lined canal (Figure 2-12). The canal is a trapezoidal, concrete-lined structure with side slopes averaging 1.8H:1V and an average bottom width of approximately 22 feet, although the cross-sectional geometry of the canal in its current condition varies greatly across the segment due to sedimentation. Based on available as-built drawings, the concrete liner is approximately 9 inches thick, with welded wire fabric reinforcement. For the upper 1,700 feet of the segment, the canal is embedded in a large, flat area of the reservoir and subject to recreational/vehicular traffic when the reservoir pool is down.



Figure 2-12. Concrete-lined Canal Downstream of Tilley No. 3 Concrete Flume, Looking Downstream (Photo Date: 02/2021)

An embankment was constructed on the left bank of the canal for most of the remainder of the segment, with rock slope protection placed on the waterside slope to protect the bank from wind and wave erosion (Figure 2-13).



Figure 2-13. Typical Embankment with Rock Slope Protection, Looking Upstream (Photo Date: 02/2021)

A 5-foot-wide Steel Bridge spans the canal just downstream of the Tilley No. 3 Concrete Flume near Station 89+00. The Steel Bridge is founded on concrete abutments at either side of the canal (Figure 2-14).



Figure 2-14. Unnamed Steel Bridge near Station 89+00 (Photo Date: 02/2021)

2.4.1.2 Refugio Concrete Box Flume

The Refugio Concrete Box Flume (Figure 2-15) is a raised concrete flume approximately 275 feet long, including transitions. The rectangular flume is 176 feet, 6 inches long, 10 feet wide, and 10 feet high. Design freeboard in the flume is approximately 1 foot, 8 inches. Fifty-foot-long concrete transitions extend on either side, connecting to the trapezoidal, concrete-lined canal. Piers range in height from 6 to 18 feet along the flume.



Figure 2-15. Refugio Concrete Box Flume (near Station 149+00), Looking Upstream (Photo Date: 02/2021)

2.4.1.3 Bailey Bridge

Bailey Bridge (Figure 2-16) is stored on Federal lands administered by the Forest Service above the high-water line, outside the Borel FERC Project boundary off Tuttle Road, adjacent to an unpaved non-Project public road. The storage area is secured with a chain-link fence and gates. The bridge was hauled to the canal when equipment was needed to cross for marina access, emergencies or operations.



Figure 2-16. Bailey Bridge, Stored on Federal Lands Administered by the Forest Service (Photo Date: 02/2021)

2.4.1.4 Access Roads

Access to the Tilley No. 3 Concrete Flume at the upstream end of **Segment 3** is described above through **Segment 2** at Station 89+00. An existing public access road on Federal lands administered by the Forest Service runs parallel to the canal and leads to the Refugio Concrete Box Flume. Existing public roads on Federal lands administered by the Forest Service at the upstream and downstream ends of the Refugio Concrete Box Flume could be used to access the lower part of **Segment 3** but are outside the Borel FERC Project boundary.

These roads extend from Sawmill Road, accessible from Access Road 174+00 (see **Segment 4** for a description). Sawmill Road is on private land and Federal lands administered by the Forest Service. Additional unpaved, non-Project access roads near the canal system on Federal lands administered by the Forest Service could be investigated for use during construction. The area on the landside of the canal is generally flat and subject to frequent recreational/vehicular traffic, and it is anticipated that the alignment of roadways will change through time depending on inundation and/or erosion.

2.4.2 SCE Proposal

The proposed actions and considerations for **Segment 3** are summarized below. Preliminary design plans reflecting the proposed action are included in Appendix C.

2.4.2.1 Decommissioning Approach

The Plan consists of the following actions:

 Concrete-lined Canal – Abandon with Modification: The existing concrete canal between STA 88+15 and 118+00 (approx. 2,985 feet long) will be backfilled with imported soil to minimize ground disturbance. The banks of the canal will remain in place and portions may be used for access. Ground disturbance will be kept to a minimum. If needed, the interior of the canal will be dewatered prior to fill placement.

- Concrete-lined Canal Demolish and Bury: The canal will be demolished and buried in place from station 118+00 to 161+00 (approx. 4,300 feet long). The upper portion of the existing concrete lining within the canal will be removed and processed (i.e., maximum particle size of 6 inches) mixed with native soil and used as backfill. Portions of the canal bank and adjacent area will be excavated to a maximum slope of 3H:1V and graded to naturally conform to the adjacent topography. Dewatering of the canal may be required prior to demolition and/or backfill.
- Steel Bridge Demolish and Haul Off Site: The Steel Bridge will be removed and recycled at an approved recycling facility.
- **Refugio Concrete Box Flume Demolish and Bury**: The Refugio Concrete Box Flume will be demolished, along with the abutments, transitions, and foundations. Piers and footings will be removed to a depth of 2 feet below existing grade. Concrete will be processed, mixed with native soil, and used as backfill in adjacent portions of the canal. Cofferdams and dewatering may be required at this location to accommodate demolition and removal of the flume. The area will be backfilled and graded to naturally conform to the adjacent topography.
- **Bailey Bridge Abandon in Place**: Bailey Bridge will no longer be needed. The canal in this area will be backfilled making it passable by pedestrians and vehicles when reservoir levels permit.

2.4.2.2 Access and Staging

To minimize ground disturbance, SCE will work with the Forest Service to access the Borel Project using roadways that are most prominent at the time of construction (Figure 2-17). These unpaved access roads will require temporary grading and other improvements for use by construction vehicles during construction. If improved, it is expected that they will be abandoned in the improved condition. Roads developed for the sole purpose of construction activities will be graded back to preconstruction conditions upon completion of work.

A 3-acre strip of land between Sawmill Road and the canal near Station 147+00 has been identified for additional access and potential staging near the Refugio Concrete Box Flume during decommissioning activities. The staging area may require a TCE as a portion of this staging area is outside the Borel FERC Project boundary and on Federal lands administered by the Forest Service. If developed, it is expected that the staging area will be graded back to pre-construction conditions upon completion of work.



Figure 2-17. Example of Recreational Roads on Federal Lands Administered by the Forest Service that may be used for Construction Access. (Photo Date: 03/2022)

2.5 Segment 4 – End of SCE Land to Auxiliary Dam (Corps)

2.5.1 Existing Features

Segment 4 extends from the end of SCE land (Station 161+00) to the Corps' Auxiliary Dam (Station 273+80), a distance of approximately 11,280 feet. Approximately 6,815 feet is Federal lands administered by the Forest Service while SCE manages approximately 5,165 feet. The following subsections describe Borel Project features within this segment.

2.5.1.1 Concrete-lined Canal

Segment 4 includes approximately 8,770 feet of concrete-lined canal (Figure 2-18). The canal is a trapezoidal, concrete-lined structure with side slopes averaging 1.8H:1V and an average bottom width of approximately 22 feet, although the cross-sectional geometry of the canal in its current condition varies greatly across the segment due to sedimentation. Based on available as-built drawings, the concrete liner is approximately 9 inches thick, with welded wire fabric reinforcement.



Figure 2-18. Concrete-lined Canal, Looking Downstream (Photo Date: 02/2021)

An embankment was constructed on the left bank of the canal for most of the segment, with rock slope protection placed on the waterside slope to protect the bank from wind and wave erosion (Figure 2-19).



Figure 2-19. Rock Slope Protection on Canal Bank (Photo Date: 02/2021)

Sawmill Bridge is a 20-foot-wide concrete bridge that crosses the canal just upstream of Rich Gulch Concrete Siphon, near Station 173+50 (Figure 2-20). The canal transitions on both sides from trapezoidal to rectangular under Sawmill Bridge.



Figure 2-20. Sawmill Bridge at Station 173+50, Looking Downstream (Photo Date: 02/2021)

2.5.1.2 Rich Gulch Concrete Siphon

Rich Gulch Concrete Siphon (Figure 2-21) is located at Station 194+00 and is approximately 510 feet long with an interior diameter of 10 feet, 6 inches. Fifty-foot-long wingwall transitions on both ends connect to the concrete-lined trapezoidal canal. Chain-link fencing has been placed on top of the entrance and exit headwalls to the siphon.


Figure 2-21. Rich Gulch Concrete Siphon Exit, Looking Upstream (Photo Date: 03/2022)

2.5.1.3 Kern River Concrete Siphon

Kern River Concrete Siphon (Figure 2-22), located at Station 210+00, is approximately 1,920 feet long with an interior diameter of 10 feet, 6 inches. Fifty-foot-long transitions on both ends connect to the concrete-lined trapezoidal canal. Chain-link fencing has been placed on top of the entrance and exit headwalls to the siphon.



Figure 2-22. Kern River Concrete Siphon Entrance, Looking Downstream (Photo Date: 03/2022)

2.5.1.4 Engineers Point

From the terminus of the Kern River Concrete Siphon, the distance between the Borel Canal to the boundary of the Corps-condemned land is approximately 5,300 feet. Fill was placed to construct the left bank of the canal for most the segment, with rock slope protection placed on the waterside slope to protect the bank from erosion (Figure 2-23).





Figure 2-23. Engineers Point, Looking Upstream (Photo Date: 02/2021)

2.5.1.5 Access Roads

Access at the upstream end of **Segment 4** is via Sawmill Road, an existing unpaved road (Access Road 174+00) that extends approximately 1 mile from SR 155 on private land and on Federal lands administered by the Forest Service. The road enters the canal between the Refugio Concrete Box Flume and Rich Gulch Concrete Siphon. A second road (Access Road 200+00) on Federal lands administered by the Forest Service extends 2,000 feet from SR 155 to the Borel Project, between the Rich Gulch and Kern River concrete siphons.

Near the Auxiliary Dam, access to Engineers Point may be limited. Corps construction activity decommissioned the primary road used to access the Borel Project. It is not yet known if the access road will be put back into service or if it will provide access to the Borel Project. When Lake Isabella water levels are low, Engineers Point can be accessed by using existing public roads on Federal lands administered by the Forest Service from the Forest Service Auxiliary Dam Campground, located on the south shore of Lake Isabella, adjacent to the Auxiliary Dam.

2.5.2 SCE Proposal

The proposed actions and considerations for **Segment 4** are summarized below. Preliminary design plans reflecting the proposed action are included in Appendix C.

2.5.2.1 Decommissioning Approach

The Plan consists of the following actions:

Concrete-lined Canal – Abandon with Modification: Two locations of existing concrete canal within this segment (between STA 168+00 and 175+00 and between STA 196+50 and 203+50) will be backfilled with imported soil to minimize ground disturbance. The banks of the canal will remain in place and portions may be used for access, but ground disturbance will be kept to a minimum. If needed, the interior of the canal will be dewatered prior to fill placement.

- **Concrete-lined Canal Demolish and Bury**: Apart from the two locations described above, the remainder of the canal within this segment will be demolished and buried in place. The upper portion of the existing concrete lining within the canal will be removed and processed to be suitable as backfill (i.e., maximum particle size of 6 inches). Portions of the canal bank and adjacent area will be excavated to a maximum slope of 3H:1V and graded to provide drainage. Where present, the left bank of the lined canal will be excavated after removal of the upper portion of the liner, and the existing rock slope protection on the waterside slope will be removed and stockpiled. The excavated native material will be blended with processed concrete liner material and used to backfill the canal. The stockpiled rock slope protection will be placed on the finished slope on the waterside to protect against wind and wave erosion.
- **Sawmill Bridge Demolish and Bury**: The concrete bridge will be demolished, and the concrete will be processed and buried with clean fill within the adjacent canal.
- **Siphons Abandon with Modification**: Rich Gulch and Kern River concrete siphons will be abandoned in place and modified with a concrete slurry/plug placed at the entry and exits. Any fencing, exposed steel, or metal safety hazards will be removed. The existing wingwalls and headwalls will remain in place. The canal will be backfilled at the entry and exits with clean fill to the top of the headwalls after the concrete slurry/plugs are constructed.

2.5.2.2 Access and Staging

A potential 0.1-acre staging area has been identified adjacent to the access road and approximately 300 feet west of the Borel FERC Project boundary at Station 200+00, between Rich Gulch and Kern River concrete siphons.

To minimize ground disturbance, SCE will work with the Forest Service for access using roadways that are most prominent at the time of construction. These unpaved access roads will require temporary grading and other improvements for use by construction vehicles during construction. If improved, it is expected that they will be abandoned in the improved condition. Roads and staging areas developed for the sole purpose of construction activities will be graded back to preconstruction conditions upon completion of work.

2.6 Segment 5 – Auxiliary Dam (Corps)

2.6.1 Existing Features

Segment 5 contains the land in the vicinity of the Auxiliary Dam, which starts at Station 273+80 and ends at Station 298+00. The lands within this segment were condemned by the Corps. An estimated 900 feet of canal upstream of the Auxiliary Dam have been filled (Figure 2-24). The canal intake works at the Auxiliary Dam have been removed, destroyed, and filled as part of seismic upgrades.



Figure 2-24. Limits of Corps Canal Backfill and Auxiliary Dam, Looking Downstream (Photo Date: 02/2021)

An estimated 800 feet of the short canal segment (below the Auxiliary Dam to the boundary of Federal lands administered by the Corps) was also filled as part of the seismic upgrade construction activities (Figure 2-25).



Figure 2-25. Federal Lands Administered by the Corps with Filled Canal and Auxiliary Dam, Looking Upstream (Photo Date: 01/2020)

2.6.2 SCE Proposal

Segment 5 is within the Federal lands administered by the Corps' and has no proposed action.

2.7 Segment 6 – Auxiliary Dam to Alta Sierra Avenue

2.7.1 Existing Features

Segment 6 extends from the end of the fill placed by the Corps as part of upgrades to the Auxiliary Dam (Station 298+00) to Alta Sierra Avenue (Station 398+48), a distance of 10,048 feet. All lands within **Segment 6** are owned by SCE. Adjacent land use ranges from light industrial to residential to vacant.

Eva Avenue, SR 178, Lake Isabella Boulevard, and Erskine Creek Road bridges all cross the canal within this segment, as does the Kern County-owned Lakeland Walk Bridge. SCE-owned distribution lines are located along the landside toe of the right bank of the canal from Lake Isabella Boulevard downstream to Golden Spur Street, where they diverge from the canal and extend to Lake Isabella Boulevard. The following subsections describe the Borel Project features within this segment.

2.7.1.1 Concrete-lined Canal

Segment 6 includes approximately 9,217 feet of concrete-lined canal (Figure 2-26). The canal is a trapezoidal, concrete-lined structure with side slopes between 1.7 and 1.9H:1V and an average bottom width of approximately 22 feet. Based on available as-built drawings, the concrete liner is approximately 9 inches thick, with welded wire fabric reinforcement.



Figure 2-26. Concrete-lined Canal, Looking Downstream (Photo Date: 02/2021)

2.7.1.2 Eva Avenue

Eva Avenue (Figure 2-27) is a two-lane, single-span, county, vehicle bridge located between the Auxiliary Dam and SR 178, near Station 303+50. Eva Avenue extends from Barlow Road on the western side of the canal and terminates at the turnaround on the eastern side of the canal before Eva Avenue reaches SR 178.



Figure 2-27. Eva Avenue (Photo Date: 03/2023)

2.7.1.3 SR 178 and Lake Isabella Boulevard Bridges

SR 178 crosses the canal near Station 309+00 via two two-lane, single-span bridges with an open median in between (Figure 2-28). Lake Isabella Boulevard is located just downstream of SR 178, near Station 310+30, and includes a two-lane, single-span, vehicle bridge (Figure 2-29). A 12-inch water line crosses under the canal at the SR 178 bridge, with an emergency 2-inch water line attached to the bridge.



Figure 2-28. SR 178 Bridges (Photo Date: 03/2022)



Figure 2-29. Lake Isabella Boulevard Bridge, Looking Upstream (Photo Date: 03/2022)

2.7.1.4 Lakeland Walk Bridge (Non-Project)

The Kern County-owned Lakeland Walk Bridge (Figure 2-30) is a concrete, single-span pedestrian bridge across the canal near Station 332+50. Chain-link fencing lines both sides of the bridge and there is vehicular gate access for SCE Borel Project operations and maintenance (O&M); however, this bridge is not part of the Borel Project. The bridge connects residential areas on both sides of the canal through a public access easement with SCE.



Figure 2-30. Lakeland Walk Bridge (Non-Project), Looking Downstream (Photo Date: 03/2022)

2.7.1.5 School House No. 1 Concrete Flume

The School House No. 1 Concrete Flume (Figure 2-31) is a rectangular, concrete flume crossing a seasonal drainage with transitions on the upstream and downstream ends to a trapezoidal canal. The flume extends from Station 349+42 to Station 349+95, a distance of 53 feet. The flume is approximately 24 feet wide and 9 feet tall. An unpaved access road runs along the toe of the flume, with drainage features to allow for flowthrough.



Figure 2-31. School House No. 1 Concrete Flume, Looking Downstream (Photo Date: 02/2021)

2.7.1.6 School House Cut

School House Cut (Figure 2-32) includes a concrete, rectangular canal cut through a bedrock outcrop from Station 360+00 to Station 367+16, a distance of 716 feet. The cut is approximately 22 feet wide and 7 feet high. The rectangular canal transitions from trapezoidal at the upstream end to a concrete flume at the downstream end. Concrete carrier beams with lateral concrete bracing have been constructed at the top of the canal to provide additional stability for a portion of the cut.



Figure 2-32. School House Cut, Looking Downstream (Photo Date: 03/2022)

2.7.1.7 School House Cut Flume

The School House Cut Flume (Figure 2-33) is a concrete flume, at the downstream end of School House Cut, that transitions to concrete-lined canal. The flume crosses a seasonal drainage and extends from Station 367+16 to Station 367+75, a distance of 59 feet. The flume is 23 feet wide and 9 feet high, with steel bracing the top of the flume laterally at 15-foot centers.



Figure 2-33. School House Cut Flume, Looking Upstream (Photo Date: 02/2021)

2.7.1.8 Erskine Creek Road Bridge

Erskine Creek Road Bridge crosses the canal via a two-lane, single-span, concrete bridge near Station 385+00 (Figure 2-34). Chain-link fencing lines the roadway on both sides, with vehicle gates on the northern side of the canal. Overhead utility lines cross the canal adjacent to the bridge.



Figure 2-34. Erskine Creek Road Bridge, Looking Downstream (Photo Date: 02/2021)

2.7.1.9 Access Roads

Access to **Segment 6** is from public roads under Kern County jurisdiction. Access to **Segment 6** at the upstream end is available via Eva Avenue (Station 303+50) from Barlow Road (a distance of approximately 4,500 feet from SR 155). Additional access will be available along a dirt road extending 500 feet from Lake Isabella Boulevard to the southern side of the Lake Isabella Boulevard Bridge (Station 310+50). Access to the downstream end of the Borel Project is available via Erskine Creek Road Bridge (Station 385+00).

2.7.2 SCE Proposal

The proposed actions and considerations for **Segment 6** are summarized below. Preliminary design plans reflecting the proposed action are included in Appendix C.

2.7.2.1 Decommissioning Approach

The Plan consists of the following actions:

- Concrete-lined Canal Demolish and Bury: The existing concrete lining within the canal will be removed and processed to be suitable as backfill (i.e., maximum particle size of 6 inches). Portions of the canal bank and adjacent area will be excavated to a maximum slope of 3H:1V and graded to provide drainage. A 14-foot-wide access road will be maintained on the right bank. Processed concrete and excavated material will be blended and placed within the canal as backfill. The bottom of the concrete liner will be pulverized prior to placement of backfill to allow for infiltration. Rock slope protection lines the right outboard slope of the canal between the pedestrian bridge and School House Cut. A portion of this material may be stockpiled for use in other locations within the Borel FERC Project boundary limits.
- Vehicle Bridges Abandon with Modifications: The canal below the vehicle bridges will be backfilled with lightweight concrete in lieu of earthen fill to minimize loading on the existing bridge foundations. The entire concrete liner will remain in place beneath the bridges to a distance of 20 feet on both sides of the bridge footings to provide protection to the canal slope. Stormwater runoff at the Lake Isabella Boulevard and Erskine Creek Road bridges currently discharge into the Borel Canal. Additional coordination will be conducted with Kern County and California Department of Transportation Stormwater staff to determine the appropriate flow control measures.
- Lakeland Walk Bridge Abandon in Place: The Kern County-owned Lakeland Walk Bridge will remain in its current condition under the existing public access easement with SCE. The canal beneath the bridge will be backfilled to match adjacent canal reaches. All existing fencing and gates will remain in place.
- School House Cut Abandon with Modifications: Fill from nearby Borel Project locations will be placed within School House Cut to the top of the rectangular channel, with a high point near the middle and the fill graded to drain to the north and south. The concrete rectangular banks of the channel will not be demolished between STA 360+00 to STA368+50. Ground disturbance in this area will be kept to a minimum.
- Flumes Demolish and Bury: The School House No. 1 Concrete Flume and School House Cut Flume will be demolished. The demolished concrete will be processed, blended with native material, and used to backfill the adjacent concrete-lined canal. Following demolition, the slopes on both sides of the flumes will be backfilled and graded to naturally conform to the adjacent topography. The rock slope protection material removed from the outside slope of the canal bank will be placed at the transitions of the canal within the existing drainage area to protect the newly constructed slopes and access road against erosion.

2.7.2.2 Access and Staging

Once the contractor has access to the canal, most of the work will be completed on SCE land within the Borel FERC Project boundary, with limited import or export of material. An existing access road parallel to the School House Cut that follows the historic alignment of the flume is outside the Borel FERC Project boundary. The road is used by SCE for regular O&M under easement with private landowners. Following construction activities, it is expected that the road will be scarified, hydroseeded, and abandoned in place.

2.8 Segment 7 – Alta Sierra Avenue to Erskine Steel Flume

2.8.1 Existing Features

Segment 7 extends from Alta Sierra Avenue (Station 398+48) to Erskine Steel Flume (Station 451+20), a distance of approximately 5,272 feet. Property ownership/management within **Segment 7** varies between private landowners and Federal lands administered by BLM. Adjacent land use within the upper portion of the segment is residential, transitioning to largely vacant parcels towards the lower end. Woodrow W. Wallace Elementary and Middle School and Kern Valley High School are located adjacent to the canal near Station 411+00. The size and orientation of private parcels varies throughout **Segment 7**. SCE-owned distribution lines that exist along the northern fence line of the corridor are expected to remain in place through existing easements with private landowners. The following subsections describe Borel Project features within this segment.

2.8.1.1 Concrete-lined Canal

Segment 7 includes approximately 5,700 feet of concrete-lined canal (Figure 2-35). The canal is a trapezoidal, concrete-lined structure with side slopes between 1.7 and 1.9H:1V and an average bottom width of approximately 22 feet. Based on available as-built drawings, the concrete liner is approximately 9 inches thick with welded wire fabric reinforcement. Utility poles are located along the landside toe of the right bank of the canal.



Figure 2-35. Concrete-lined Canal, Looking Downstream (Photo Date: 02/2021)

2.8.1.2 School Pedestrian Bridge

The School Pedestrian Bridge (Figure 2-36) is located at Webb Avenue. It is a steel, single-span pedestrian bridge that crosses the canal near Station 413+50 and connects residential areas on the western side of the canal to the middle school and high school on the eastern side of the canal. The bridge is approximately 4 feet wide and surrounded by chain-link material, and chain-link fencing extends across the canal bank on either side. The bridge abutments have been constructed behind the top of the concrete liner.



Figure 2-36. School Pedestrian Bridge, Looking Downstream (Photo Date: 01/2020)

2.8.1.3 County Emergency Vehicular Bridges (Non-Project)

Kern County has two adjacent parallel non-Project emergency vehicular bridges that cross the canal near Station 428+75 (Figure 2-37). These single-span bridges have concrete decks and steel supporting members. Concrete abutments have been constructed at the top of the canal bank on both sides for each bridge.



Figure 2-37. County Emergency Vehicular Bridges (Non-Project) (Photo Date: 01/2020)

2.8.1.4 Erskine Steel Flume

The Erskine Steel Flume (Figure 2-38) is a steel Lennon flume with steel support braces. The flume extends from Station 442+25 to Station 451+20, a distance of 895 feet. A steel plank walkway is located along the flume for pedestrian access. Metal chain-link fencing and gates are on both sides of the flume and limit access to the canal from the bridge.



Figure 2-38. Erskine Creek Flume, Looking Upstream (Photo Date: 01/2020)

2.8.1.5 Access Roads

Access to **Segment 7** is via public roads under Kern County jurisdiction. Access at the upstream end of **Segment 7** is available via Erskine Creek Road. Access at the downstream end of the segment is available via Commercial Avenue, an existing unpaved road that enters the canal within the limits of the Erskine Steel Flume near Station 444+00. The road is regularly used by the public and crosses under the Erskine Steel Flume. Additional access may be available along the segment if required, for example, at the County Emergency Vehicular Bridges (non-Project).

2.8.2 SCE Proposal

The proposed actions and considerations for **Segment 7** are summarized below. Preliminary design plans reflecting the proposed action are included in Appendix C.

2.8.2.1 Decommissioning Approach

The Plan consists of the following actions:

 Concrete-lined Canal – Demolish and Bury: The existing concrete lining within the canal will be removed and processed to be suitable as backfill (i.e., maximum particle size of 6 inches). Portions of the canal bank and adjacent area will be excavated to a maximum slope of 3H:1V and graded to provide drainage. The bottom of the concrete liner within BLM-managed parcels will be pulverized prior to placement of backfill to allow for infiltration.

- **Concrete-lined Canal Demolish and Haul Off Site**: The entire concrete lining will be removed from the canal within private parcels. This concrete will be spread out and buried in the backfill of adjacent reaches of the canal that are owned by SCE.
- School Pedestrian Bridge Demolish and Haul Off Site: The pedestrian bridge will be removed and will be recycled at an approved recycling facility. The bridge abutments will be removed and hauled away.
- County (Non-Project) Emergency Vehicular Bridges Abandon in Place: The non-Project County Emergency Vehicular Bridges will remain in place. Beneath the County Emergency Vehicular Bridges, the canal will be backfilled with lightweight concrete fill to match the grading in adjacent canal reaches. All fencing and gates will remain in place. The bridges will be protected in place and remain operational after decommissioning activities.
- Erskine Steel Flume Demolish and Haul Off Site: Erskine Steel Flume will be demolished, and the materials will be hauled off site. Structural steel and steel sheeting materials will be recycled at an approved recycling facility. The foundations and piers will be removed to existing grade and hauled away. The areas affected by decommissioning activities will be graded to conform to adjacent non-affected areas and hydroseeded and planted to restore natural vegetation and prevent erosion.

2.8.2.2 Access and Staging

The one unpaved portion of Commercial Avenue used to access the Borel Project at Erskine Steel Flume may be upgraded for construction purposes. It is expected that the road will be left in an improved condition upon completion of work.

2.9 Segment 8 – Erskine Steel Flume to Bodfish Siphon

2.9.1 Existing Features

Segment 8 extends from the end of the Erskine Steel Flume (Station 451+20) to Bodfish Siphon (Station 506+29), a distance of approximately 5,509 feet. Property management/ownership within **Segment 8** includes privately held parcels, SCE-owned land, and Federal land administered by BLM. The following subsections describe the Borel Project features within this segment.

2.9.1.1 Concrete-lined Canal

Segment 8 includes approximately 4,880 feet of concrete-lined canal (Figure 2-39). The canal is a trapezoidal, concrete-lined structure with side slopes between 1.7 and 1.9H:1V and an average bottom width of approximately 22 feet. Based on available as-built drawings, the concrete liner is approximately 9 inches thick with welded wire fabric reinforcement. In this area, the canal extends through a portion of the Borel Project called the Highbank, and includes private parcels, Federal land administered by BLM, and SCE land.



Figure 2-39. Concrete-lined Canal, Looking Downstream (Photo Date: 01/2020)

2.9.1.2 Bodfish Siphon

Bodfish Siphon is approximately 629 feet long with an interior diameter of 10 feet, 6 inches (Figure 2-40). The siphon crosses underneath Lake Isabella Boulevard and Bodfish Creek. Transitions on both ends connect to the concrete-lined trapezoidal canal. Chain-link fencing has been placed on top of the entrance and exit headwalls to the siphon. A gauging station is located at the siphon inlet and vaults and valves located at the low point of the siphon accommodate dewatering of the siphon and discharge into Bodfish Creek.



Figure 2-40. Bodfish Siphon, Looking Downstream (Photo Date: 01/2020)

2.9.1.3 Access Roads

Access to the upstream end of **Segment 8** is via Commercial Avenue, an existing unpaved road under Kern County jurisdiction that enters the canal within the limits of Erskine Steel Flume near Station 444+00 as described above for **Segment 7**. Additional access to **Segment 8** is from Canal Road, which splits and provides access the canal in two locations. The roads are under Kern County jurisdiction and on Federal land administered by the BLM. Only one road is regularly used to access the canal at Station 477+00. The older unused access road accessed the canal at Station 465+00.

2.9.2 SCE Proposal

The proposed actions and considerations for **Segment 8** are summarized below. Preliminary design plans reflecting the proposed action are included in Appendix C.

2.9.2.1 Decommissioning Approach

The Plan consists of the following actions:

- **Concrete-lined Canal Demolish and Bury:** Within private parcels, the concrete liner will be demolished and processed (i.e., maximum particle size of 6 inches) and be used as backfill. The land will be graded to collect offsite drainage and direct stormwater flows toward Erskine Creek away from existing structures located outside the Borel FERC Project boundary.
- Concrete-lined Canal Demolish and Bury: Within SCE lands and lands administered by the BLM, the upper portion of the existing concrete lining materials within the canal will be removed and processed to be suitable as backfill (i.e., maximum particle size of 6 inches). Portions of the canal bank and adjacent area will be excavated to a maximum slope of 3H:1V and graded to provide drainage. A 14-foot-wide access road will be maintained on the right bank. Processed concrete and excavated native material will be blended and placed within the canal as backfill. The bottom of the concrete liner will be pulverized prior to placement of backfill to allow for infiltration. This area of the segment will be graded to drain toward Bodfish Siphon.
- Bodfish Siphon Abandon with Modification: Bodfish Siphon will be abandoned in place with modifications. The wingwalls will be demolished, and the headwalls will be protected in place. The siphon will be filled with concrete slurry, and the gauging station and concrete vaults used to dewater the siphon will be removed and filled in. The entrance of the siphon will be plugged. The land around the siphon entrance and headwalls will be graded to accommodate drainage from the Highbank area. A trapezoidal channel that is lined with rock slope protection, will convey stormwater runoff down the slope and dissipate energy prior to discharge into a biofiltration basin that will be located at the northwestern corner of the Lake Isabella Boulevard and Bodfish Canyon Road intersection (Figure 2-41). The existing culvert under Lake Isabella Boulevard will be replaced with a larger culvert to accommodate stormwater flows that exceed the capacity of the biofiltration basin. The culvert will discharge into a subsequent biofiltration swale on the SCE property (east of Lake Isabella Boulevard). During large storm events, the swale will convey stormwater runoff into Bodfish Creek.



Figure 2-41. Location of Proposed Biofiltration Basin at Bodfish Siphon (Photo Date: 03/2022)

2.9.2.2 Access and Staging

Primary access to the canal is anticipated to be from the access road at the upstream end of **Segment 8** via Commercial Avenue and the road accessing the canal at Station 477+00. Road Station 465+00 is not intended to be used for construction to reduce impacts on local residents. It is expected that if access roads are improved for construction, they will be left in an improved condition upon completion of work. A 3.0-acre parcel adjacent to Bodfish Canyon Road will be utilized for staging of construction trailers only. Access to the staging area would be via Lake Isabella Boulevard and/or Bodfish Canyon Road at Station 503+00.

2.10 Segment 9 – Bodfish Siphon to Pioneer Steel Siphon

2.10.1 Existing Features

Segment 9 extends from the end of Bodfish Siphon (Station 506+29) to Pioneer Steel Siphon (Station 548+79), a distance of approximately 4,250 feet. Most land within **Segment 9** is administered by BLM, except for the canal segment upstream of Tunnel No. 1 (SCE) and the last 240 feet of Pioneer Steel Siphon (Federal lands administered by the Forest Service). A communication line owned by Frontier runs parallel to the canal within the Borel FERC Project boundary. The following subsections describe Borel Project features within this segment.

2.10.1.1 Concrete-lined Canal

Segment 9 includes approximately 3,150 feet of concrete-lined canal (Figure 2-42). The canal is a trapezoidal, concrete-lined structure with side slopes between 1H:2V and 1H:4V and an average bottom width of approximately 15 feet. Based on available as-built drawings, the concrete liner is approximately 9 inches thick, with welded wire fabric reinforcement. An emergency overflow release is located just downstream of Tunnel No. 1 at Station 520+00.



Figure 2-42. Concrete-lined Canal, Looking Upstream (Photo Date: 03/2022)

A retaining wall is located on the right bank slope downstream of Tunnel No. 1 from approximately Station 526+80 to Station 528+00 (Figure 2-43). The wall includes vertical H piles with corrugated metal lagging. Drain lines are on either end of the wall. The depth of the H piles is unknown.



Figure 2-43. Retaining Wall Structure Downstream of Tunnel No. 1, Looking Downstream (Photo Date: 01/2020)

2.10.1.2 Tunnel No. 1

Tunnel No. 1 (Figure 2-44) extends from Station 512+30 to Station 519+30, a distance of 700 feet. The tunnel is horseshoe shaped with a maximum height of 12 feet and a bottom width of 10 feet. An access road provides access to the tunnel entrance and exit.



Figure 2-44. Tunnel No. 1 Entrance, Looking Downstream (Photo Date: 06/2020)

2.10.1.3 Pioneer Steel Siphon

Pioneer Steel Siphon is approximately 400 feet long with an interior diameter of 10 feet, 6 inches (Figure 2-45). The siphon crosses a gully to connect two sections of canal. A gauge station structure with power supplied via overhead line is located on the north side of the siphon. Concrete headwalls transition to the concrete-lined trapezoidal canal.



Figure 2-45. Pioneer Steel Siphon, Looking Upstream (Photo Date: 02/2021)

A concrete headwall directs stormwater drainage from the tributary area upstream of the siphon into three 36-inch corrugated steel pipe culverts that convey stormwater downstream of the siphon. Other appurtenant structures include valves, drainpipe, and concrete box to collect and dissipate energy from discharge flows when the siphon is dewatered (Figure 2-46).



Figure 2-46. Pioneer Steel Siphon Dewatering and Drainage Features (Photo Date: 02/2021)

2.10.1.4 Access Roads

Access to the canal in **Segment 9** is from roads on Federal lands administered by the BLM and Forest Service. At the upstream end of the segment, an unpaved public road on Federal lands administered by the BLM is accessed through a residential neighborhood via Miller Street, with an entrance approximately 1,500 feet from Lake Isabella Boulevard. The road enters the canal near Tunnel No. 1, with three access points (Station 512+00, Station 514+50, and Station 519+50). At the downstream end of the segment near the Pioneer Siphon, access is via an unpaved public road on Federal lands managed by the Forest Service and the BLM. The road enters the system at Pioneer Siphon near Station 549+00, extending 950 feet from Kern River Canyon Road. This access road continues to the upstream side of Pioneer Steel Siphon and accesses the system near Station 537+00. An area of 3.08 acres between the canal and Access Road Station 537+00 will be used for staging and access.

2.10.2 SCE Proposal

The proposed actions and considerations for **Segment 9** are summarized below. Preliminary design plans reflecting the proposed action are included in Appendix C.

2.10.2.1 Decommissioning Approach

The Plan consists of the following actions:

• **Concrete-lined Canal – Demolish and Bury**: The upper portion of the concrete liner will be removed to a depth of approximately 1 foot below finished grade on both sides of the canal and processed to be suitable as backfill (i.e., maximum particle size of 6 inches). The native soils that compose the right bank will be excavated and blended with the processed concrete for use as fill material. The excavation of the right bank will lower the ground elevation at the retaining wall, and the steel H beams and wood bulkheads will be removed and hauled away. The fill material will be placed in the remaining portion of the canal to provide positive drainage across the Borel FERC Project boundary. The Frontier-owned communication line is expected to be removed and the access road abandoned as part of the grading activities. After mass grading activities are completed, this area will be hydroseeded for erosion control.

• **Tunnel No. 1 – Abandon with Modification:** The tunnel will be backfilled with a mixture of concrete debris and flowable fill material. Work will need to be done in compliance with Occupational Safety and Health Administration Tunnel Safety Orders. In compliance with these orders, ventilation for the tunnel and additional tunnel utilities, such as lighting, will be installed. A track loader will move the lower tunnel fill materials to the midway point from both portals.

After the lower fill procedure is complete, the upper half of the tunnel will be filled with concrete slurry. Concrete pumping pipes will be placed in the tunnel, and forms will be set at the portal entrances. These forms will have access doors to allow personnel to safely enter and exit the tunnel. At the midway point in the tunnel, the concrete pumping pipe will be equipped with an air slugger. This air slugger will inject compressed air at the discharge area to help compact the concrete slurry to completely fill the tunnel. As the tunnel is filled, ventilation duct/lighting and concrete pumping pipe sections will be removed, allowing the concrete slurry placement to proceed to the portal areas. Near the ends, the access doors in the forms will be tightly secured, allowing concrete slurry to completely fill the tunnel. The headwalls will remain, and the area will be backfilled and graded to provide a surface that naturally conforms to the adjacent topography prior to seeding the disturbed areas.

• Pioneer Steel Siphon – Demolish and Haul Off Site: Based on visual inspection, it appears that the steel pipe coating contains lead. Any lead remediation will be determined following testing and verification of the levels of contaminants present. If necessary, the coatings will be removed by an abatement contractor prior to starting demolition of the pipe. After coatings removal, the pipe will be cut into manageable lengths and loaded onto trucks for transport to a recycling facility. The concrete footings will be left in place. Appurtenant facilities, including the gauging station building and siphon, drainpipe, and concrete energy dissipation structure, will be removed and hauled away. The storm drainage head wall culverts will be protected in place, and rip rap will be placed along the embankment at the inlet and at the pipe discharge points to minimize erosion. The overhead communications and power lines are expected to be removed.

2.10.2.2 Access and Staging

The existing unpaved roads used to access the features in this segment may require temporary grading and other improvements for use by construction vehicles during construction. If improved, it expected that they will be abandoned in the improved condition.

2.11 Segment 10 – Pioneer Steel Siphon to Forebay Structure

2.11.1 Existing Features

Segment 10 extends from the end of Pioneer Steel Siphon (Station 548+79) to the Forebay Structure (Station 628+74), a distance of approximately 7,995 feet. All lands within **Segment 10** are Federally owned and administered by the Forest Service). A communication line owned by Frontier runs parallel to the canal within the Borel FERC Project boundary. The following subsections describe Borel Project features within this segment.

2.11.1.1 Concrete-lined Canal

Segment 10 includes approximately 6,053 feet of concrete-lined canal (Figure 2-47). The canal is a trapezoidal, concrete-lined structure with side slopes between 1.2 and 1.4H:1V and an average bottom width of approximately 15 feet. Based on available as-built drawings, the concrete liner is approximately 9 inches thick, with welded wire fabric reinforcement.



Figure 2-47. Concrete-lined Canal, Looking Downstream of Tunnel No. 1 1/2 (Photo Date: 02/2021)

2.11.1.2 Tunnel No. 1 1/2

Tunnel No. 1 1/2 (Figure 2-48) extends from Station 566+35 to Station 571+05, a distance of 470 feet. The tunnel is horseshoe-shaped, with a maximum height of 12 feet and a bottom width of 10 feet. An access road extends to the entry and exit portals to the tunnel. The access road is relatively steep and may limit the type of construction equipment that can access this area.



Figure 2-48. Tunnel No. 1 1/2 Entrance, Looking Downstream (Photo Date: 02/2021)

2.11.1.3 Tunnel No. 2

Tunnel No. 2 (Figure 2-49) extends from Station 581+75 to Station 587+50, a distance of 575 feet. The tunnel is horseshoe-shaped, with a maximum height of 12 feet and a width of 10 feet. An access road provides access to the tunnel entrance and exit.



Figure 2-49. Tunnel No. 2 Entrance, Looking Downstream (Photo Date: 02/2021)

Utility poles line the right bank of the canal downstream of Tunnel No. 2 (Figure 2-50).



Figure 2-50. Utility Poles on Right Bank Downstream of Tunnel No. 2, Looking Downstream (Photo Date: 02/2021)

2.11.1.4 Flume No. 623

Flume No. 623 is a steel Lennon flume with steel support braces (Figure 2-51). The flume extends from Station 596+00 to Station 597+23, a distance of 123 feet. A wooden plank walkway is located

along the center of the flume for pedestrian access. Chain-link fencing and gates are located at both ends of the flume to limit access.



Figure 2-51. Flume No. 623, Looking Downstream (Photo Date: 02/2021)

2.11.1.5 Tunnel No. 3

Tunnel No. 3 (Figure 2-52) extends from Station 613+74 to Station 615+81, a distance of 207 feet, crossing under Borel Road near the intersection with Kern River Canyon Road. The tunnel is horseshoe-shaped, with a maximum height of 12 feet and a bottom width of 10 feet. Access roads extend to the entry and exit portals of the tunnel from Borel Road.



Figure 2-52. Tunnel No. 3 Exit, Looking Upstream (Photo Date: 02/2021)

2.11.1.6 Profanity Steel Flume

Profanity Steel Flume (Figure 2-53) is a steel Lennon flume with steel support braces. The flume extends from Station 623+07 to Station 625+47, a distance of 240 feet. A wooden plank walkway is located along the center of the flume for pedestrian access. Metal chain-link fencing and gates are on either end of the flume to limit access.



Figure 2-53. Profanity Steel Flume, Looking Downstream (Photo Date: 02/2021)

2.11.1.7 Access Roads

Several roads on Federal land administered by the Forest Service provide access to the canal in **Segment 10**. Access at the upstream end of the segment is through an unpaved public access road (Access Road Station 549+00), described above for **Segment 9**. At the downstream end, an unpaved public access road near Station 585+00 and Tunnel No. 2 extends approximately 1,400 feet from Kern River Canyon Road. Borel Road connects SR 178 and Kern River Canyon Road, crossing the system at Tunnel No. 3. Mid-segment, there are two gated road loops located near Tunnel No. 1 1/2 and Tunnel 2 that provide SCE Borel Project access around steep terrain.

2.11.2 SCE Proposal

The proposed actions and considerations for **Segment 10** are summarized below. Preliminary design plans reflecting the proposed action are included in Appendix C.

2.11.2.1 Decommissioning Approach

The Plan consists of the following actions:

- **Concrete-lined Canal Demolish and Bury**: The concrete liner will be removed to a depth of approximately 1 foot below finished grade on both sides of the canal, and the materials will be processed to be suitable as backfill (i.e., maximum particle size of 6 inches). The right bank will be excavated, and the materials will be blended with the processed concrete for use as fill material. The Frontier-owned communication line is expected to be removed, and the access road abandoned as part of the grading activities. The fill material will be placed in the remaining portion of the canal to provide positive drainage, and the area will be covered with hydroseed for erosion control after decommissioning activities are completed.
- **Tunnels No. 1 1/2, 2, and 3 Abandon with Modification:** The tunnels will be backfilled with a mixture of concrete debris and flowable fill material. Work be done in compliance with Occupational Safety and Health Administration Tunnel Safety Orders. In compliance with these orders, ventilation for the tunnel and additional tunnel utilities, such as lighting, will be installed. A track loader will move the lower tunnel fill materials to the midway point from both portals.

After the lower fill procedure is complete, the upper half of the tunnels will be filled with concrete slurry. Concrete pumping pipes will be placed in the tunnel, and forms will be set at the portal entrances. These forms will have access doors to allow personnel to safely enter and exit the tunnel. At the midway point in the tunnel, the concrete pumping pipe will be equipped with an air slugger. This air slugger will inject compressed air at the discharge area to help compact the concrete slurry to completely fill the tunnel. As the tunnel is filled, ventilation duct/lighting and concrete pumping pipe sections will be removed, allowing the concrete slurry placement to proceed to the portal areas. Near the ends, the access doors in the forms will be tightly secured, allowing concrete slurry to completely fill the tunnel. The headwalls will remain, and the area will be backfilled and graded to provide a surface that naturally conforms to the adjacent topography prior to seeding the disturbed areas.

• Flume No. 623 and Profanity Steel Flume – Demolish and Haul Off Site: Flume No. 623 and Profanity Flume will be demolished, and the materials will be hauled off site. To minimize fire hazards, demolition of the steel flumes will be completed by dismantling instead of flame cutting. Dismantling would begin by dropping the Lennon sheets and hanger/compression rods to the ground, where these materials will be collected and placed into either a truck or dumpster for hauling to a recycling facility. Working from lifts, the crew will remove bolted connections on the substructure steel.

A 50-ton crane (or similar) will assist in removing the substructure steel materials by lifting these components away from the footings. The substructure steel materials will also be loaded onto trucks for hauling to a recycling facility. Existing concrete footings will remain in place to minimize ground disturbance.

2.11.2.2 Access and Staging

The gated road loop located above Tunnel No. 1 1/2 (between STA 566+00 to 571+00) is steep and narrow. Grading in this area will not be allowed and access across this point will be limited, and large equipment will require transport around this area using Kern River Canyon Road and the access road near STA 585+00. Upon completion of construction, the gated road loops will be scarified (surficial), hydroseeded, and abandoned.

The primary unpaved access roads connecting to the Kern River Canyon Road may require temporary grading and other improvements for use by construction vehicles during construction. If improved, it is expected that they will be abandoned in the improved condition.

A 0.21-acre staging area which may extend beyond the Borel FERC Project boundary has been identified adjacent to Tunnel No. 3 and Borel Road, near Station 615+00. If developed, it is expected that the staging area will require a TCE and will be graded back to pre-construction conditions upon completion of work.

2.12 Segment 11 – Forebay Structure to Tailrace

2.12.1 Existing Features

Segment 11 extends from the Forebay Structure to the Powerhouse and Tailrace. The following subsections describe Borel Project features within this segment. This segment of the Borel Project

does not follow the Borel Project control line. Therefore, unlike the previous segment descriptions, stationing is not provided for these facilities.

2.12.1.1 Forebay Structure

The Forebay Structure includes a trapezoidal, concrete-lined forebay leading to a trash rake structure that removes debris from the flow prior to discharge into the penstocks (Figure 2-54). Chain-link fencing lines the top of the concrete lining and top of the bank near the county road.



Figure 2-54. Forebay Structure (Photo Date: 02/2021)

2.12.1.2 Penstocks

From the Forebay Structure, four riveted-steel penstocks (Figure 2-55) convey water to the Powerhouse main turbines. Penstock No. 1 is approximately 526 feet long and has an inner diameter of 60 inches. The steel pipe feeds Unit No. 1 within the Powerhouse. Penstock No. 2 is approximately 565 feet long, with an inner diameter of 60 inches, and feeds Unit No. 2. Penstocks No. 3 and No. 4 have an inner diameter of 60 inches for 622 feet before they join one 94-foot-long, 84-inch-inner-diameter pipe that supplies Unit No. 3.



Figure 2-55. Penstocks (Photo Date: 03/2022)

2.12.1.3 Borel Powerhouse

The Powerhouse building (Figure 2-56) is approximately 168 feet long by 64 feet wide and is constructed of reinforced concrete. The control room is located on the ground floor of the structure and houses the control equipment. A machine shop area is located on the generator floor of the Powerhouse. A separate lunch/meeting room is located on the ground floor of the Powerhouse.



Figure 2-56. Powerhouse Interior (Photo Date: 01/2020)

The Powerhouse contains two main horizontal-shaft, Francis-type, reaction turbines directconnected to Bullock generators and one main vertical-shaft, Francis-type, reaction turbine directconnected to a General Electric generator. The total rated horsepower (HP) is 15,800 HP.

The Unit No. 1 and No. 2 main turbines are directly connected to horizontal shaft, partially enclosed generators, with cooling provided by air drawn from within the Powerhouse. Unit No. 3 is a vertical shaft, fully enclosed generator, with cooling provided by once-through air drawn from inside the Powerhouse and passed through a humidifier. The Powerhouse total installed capacity is 12,000 kW.

2.12.1.4 Powerhouse Auxiliary Buildings

A corrugated metal and wooden frame storage building is located to the west of the Powerhouse (Figure 2-57). It currently contains wood planks and an assortment of small Powerhouse parts.



Figure 2-57. Storage Building (Photo Date: 03/2022)

A wooden pump house shed containing a small pump is located to the west of the Powerhouse (Figure 2-58). The pump was lowered into the Kern River using a rail guide and used to provide water to the Powerhouse operations when needed.



Figure 2-58. Pump House (Photo Date: 03/2022)

A vault toilet restroom/outbuilding made of cinder blocks is located in the Powerhouse yard at the north edge of the parking lot (Figure 2-59). The electrical supply is connected to the Powerhouse underneath the parking lot asphalt.



Figure 2-59. Restroom/Outbuilding at Powerhouse (Photo Date: 03/2022)

2.12.1.5 Non-Project Switchyard

The non-Project Borel Switchyard (Switchyard) is located adjacent to the Powerhouse (Figure 2-60). It is within the Borel FERC Project boundary, but under separate easement with the Forest Service. The Switchyard will remain and be protected in place given its importance to system reliability and protection. The main transformer bank is connected to the 66 kV bus (non-Project) through a 3-pole, 800-amp, 66-kV oil circuit breaker. Associated disconnect switches, grounding switches, potential devices, and other related equipment are located in the Switchyard. Some of the controls for the Switchyard are located inside the Powerhouse and will be moved within the Switchyard footprint.



Figure 2-60. Borel Switchyard and Powerhouse (Photo Date: 03/2022)

2.12.1.6 Tailrace

The tailrace consists of a 65-foot-long tunnel extending from the Powerhouse to the Kern River (Figure 2-61). The spillway tunnel is an arch tunnel with a width of 16 feet and maximum height of 24 feet.



Figure 2-61. Tailrace Outlet at Kern River (Photo Date: 03/2022)

2.12.1.7 Spillway Channel

A natural spillway channel extends from the end of the Forebay Structure to the Kern River. The channel was used by SCE when a Borel Project outage or O&M required dewatering of the canal and Forebay Structure.

2.12.1.8 Access Roads

Several roads in **Segment 11** are used to access Borel Project facilities. All are on lands administered by the Forest Service within the Borel FERC Project boundary. There is one primary paved road that leads to the Powerhouse, Switchyard, and appurtenant facilities from Borel Road. Multiple unpaved roads provide access to Profanity Steel Flume, the Penstocks, and non-Project distribution and transmission lines.

2.12.2 SCE Proposal

The proposed actions and considerations for **Segment 11** are summarized below. Preliminary design plans reflecting the proposed action are included in Appendix C.

2.12.2.1 Decommissioning Approach

The Plan consists of the following actions:

• Forebay Structure – Demolish and Haul Off Site: Due to the limited access (i.e., very steep and small footprint) to the Forebay Structure, a 250-ton crane (or similar) will be set up on the wide section of Kern River Canyon Road above the Forebay. This crane will encroach into one lane of Kern River Canyon Road, and a traffic control plan will be implemented.

Prior to concrete demolition, a crew will remove all existing electrical and lighting equipment, followed by removal of the existing Atlas Polar trash rake/conveyor belt system and gates. Miscellaneous mechanical removal and demolition includes the control building, stairs, handrails, intake screens, deck plates, skimmer plates, cabinets, and other items.

Concrete debris not used for backfilling will be loaded into skip boxes, lifted out by the crane, and hauled to a section of the canal where it can be processed and mixed for use as material. Revegetation plans for the forebay area will be developed during the completion of design.

 Penstocks – Demolish and Haul Off Site: The penstocks and anchor blocks will be removed and hauled off site. The concrete footings will be removed to existing grade to minimize ground disturbance and maintain slope stability. Demolition activities will include access improvements above and below the existing access road, near the mid-span of the Penstocks. Some large trees as well as other vegetation along the penstock route will need to be trimmed back or removed for the road extensions and/or fire protection.

A fire protection system will be required for this operation that will include a large water tank on Kern River Canyon Road, above the existing Forebay Structure, with polyvinyl chloride piping and manifolds to the Powerhouse. The tank will be filled via water trucks.

The steel penstock pipe will be cut into 10-foot to 20-foot-long sections and will be hauled off site to a recycling facility. The portions of the concrete anchor blocks and footings will be hauled off site to a waste facility. A crane will be staged at three or four locations along the alignment to remove the pipe segments. Revegetation plans for the penstock area will be developed during the completion of design.

• Borel Powerhouse – Demolish and Haul Off Site: Prior to construction, the Powerhouse will be tested for hazardous materials (e.g., lead and asbestos). Once any contaminants are identified and concentrations known, a hazardous material abatement plan will be developed as needed. The concentrations dictate the disposal process, and it is anticipated that a demolition notification (that states whether there is lead and asbestos in the Powerhouse materials) will be sent to the South Coast Air Quality Management District and a Building and Safety Permit will be required.

The proposed approach to demolishing the Powerhouse will begin with removing the turbines, transformers, valves, and heavy equipment inside the Powerhouse using the existing 40-ton overhead crane (Figure 2-62). Once the equipment has been removed, work on plugging the draft tubes and generator pits will be filled with concrete slurry to the final ground grade.

Work on demolition of the Powerhouse structure will require installation of scaffolding on all sides of the Powerhouse's exterior walls. Because the existing Switchyard adjacent to the Powerhouse is to remain intact, the scaffolding will function as a protective barrier between the northern Powerhouse wall and the Switchyard. A full lock-out/tag-out outage on the Switchyard will occur prior to scaffold installation.

The roof and rafters will be removed first. Lifts operated from inside the Powerhouse will perform most of the demolition. These lifts will be supported by crew on scaffolding on the outside of the Powerhouse and a crane. After the roof and rafter removal procedure is complete, the 40-ton overhead crane (in the Powerhouse; Figure 2-62) will be removed. Lastly, the concrete walls will be demolished, leaving only the foundation in place.



Figure 2-62. Existing Crane (Photo Date: 01/2020)

- **Non-Project Switchyard –** There is no proposed action for the Switchyard because it is vital to SCE's power distribution system reliability and protection.
- **Powerhouse Auxiliary Buildings Demolish and Haul Off Site:** The storage building, pump house, and restroom will be removed and demolished (including slab/footing) to grade and hauled away. The contents of the buildings will be removed and disposed of off-site prior to demolition. Ground disturbed by construction activities will be graded and hydroseeded.
- **Tailrace Demolish and Bury**: The Tailrace tunnel will be demolished and backfilled with a blend of native material and processed concrete for slope stability. After grading, clean riprap will be placed on the finished surface at a slope that conforms to the contours of the Kern River channel bank. The riprap will serve to protect the slope from erosion during high flows. Depending on the Kern River water level at the time of construction, dewatering of the tailrace may be required.
- **Spillway –** There is no proposed action for the Spillway as it is very steep, dominated by bedrock, and contains pockets of naturally occurring vegetation.

2.12.2.2 Access and Staging

There is a paved access road from Borel Road to the Powerhouse and auxiliary buildings (near Station 628+74) (Figure 2-63). Access to the Penstocks is available via unpaved existing access roads on Federal lands administered by the Forest Service from Borel Road (near Station 615+00), which connects directly to SR 178. There is paved access from Borel Road to the Powerhouse and auxiliary buildings (near Station 628+74) (Figure 2-63). A staging area (1.10 acres) has been identified adjacent to the Borel FERC Project boundary near the Powerhouse entrance (Figure 2-64). Once construction is complete, the paved access road to the Powerhouse site will be left in place while it is expected that the remaining unpaved access roads and staging area in the segment will be graded, scarified and hydroseeded unless otherwise required for SCE access to non-Project distribution or transmission lines.



Figure 2-63. Powerhouse Access Road (Photo Date: 03/2022)



Figure 2-64. Staging Area for Penstock Demolition (Mid-slope) (Photo Date: 03/2022)

3 Estimated Schedule

SCE developed an estimated schedule for the Plan and can be viewed in Appendix D – Estimated Schedule. The schedule outlines anticipated activities and estimated durations for 11 phases.

Project Management and Stakeholder Outreach phases will continue through the entire schedule.

Application Development, Stakeholder Review and File Surrender Application phases are complete. These included the development of the Surrender Application and consultation with Stakeholders, as well as the 50-day stakeholder review of the draft Surrender Application and SCE filing of the Surrender Application with FERC on May 1, 2023.

The subsequent six phases are SCE's best estimate of the schedule with acknowledgement that the timing and duration of each phase is largely outside of SCE's control.

The FERC Review Phase is expected to include several elements including FERC's Notice of Filing and request for public comment, any Additional Information Requests (AIRs), the Ready for Environmental Assessment Notice, the Draft Environmental Assessment (EA), continued ESA Section 7 and NRHA Section 106 consultations and Final EA. The FERC Review Phase will conclude when FERC issues its order approving license surrender and notifies SCE that decommissioning activities may commence.

The SWRCB Water Quality Certification Phase assumes SCE will submit to the SWRCB in 2023, an application for CWA Section 401 water quality certification to support FERC's issuance of a surrender order. SCE assumes the SWRCB will within one year, issue a water quality certification or otherwise address SCE's application, including complying, as appropriate, with the California Environmental Quality Act (CEQA).

The Design and Permitting Phase, which is estimated to take approximately two to three years, may begin prior to FERC's notice that decommissioning activities may commence but will not be completed until such time as the Final EA is issued, ESA Section 7 and NHPA Section 106 consultations are completed and CEQA and water quality certification have been completed. The Design and Permitting Phase includes refinement of the current 30% design plans, development of the 60%, 90% and final construction design packages as well as continued coordination with permitting agencies, utilities and landowners directly impacted by future construction activities. As designs are being finalized (i.e., 60% stage), SCE will apply for additional required permits, such as a Corps' Section 404 permit for impacts to jurisdictional aquatic resources.

The Supplemental Investigations Phase is expected to occur during the FERC Review Phase, SWRCB Water Quality Certification Phase and Design and Permitting Phase. The nature and duration of these investigations are dependent on the specific information required by FERC, SWRCB, State and Federal Agencies, private landowners and additionally by SCE as it continues to develop and refine the decommissioning engineering designs, specifications and supporting measures.

The Construction Phase will begin when the Design and Permitting Phase is complete. Construction is estimated to take approximately three years and will be phased to the extent possible to minimize

impacts to the community and environment. In an effort to meet these objectives, a construction plan will be developed in advance of design completion and will be communicated to Stakeholders. To expedite construction activities throughout the Borel Project area, SCE expects to have multiple construction crews working simultaneously on Borel Project segments.

The Post Construction Monitoring Phase will include resource monitoring as prescribed by the proposed measures outlined in Section 4 below and in accordance with applicable permits. Where required, post construction monitoring is expected to commence once construction and restoration activities within a Borel Project segment are completed, though the specifics of duration and timing will vary by segment and landowner.
4 Proposed Measures

SCE proposes to include a comprehensive suite of measures as part of the proposed action to ensure appropriate resource protections during Borel Project decommissioning. Table 4-1 summarizes the measures that are proposed for protection of the human and natural environment.

Table 4-1. Measures for	[·] Borel Proje	ect Decommissioning
-------------------------	--------------------------	---------------------

Measure	Subject	Measure	Applicable Segment
General Cor	struction Measures		
1	Permits	SCE will consult with the applicable Federal, State, and local agencies to obtain necessary permits and will comply with these permits during all decommissioning activities.	All
2	Borel Project Footprint	Work area footprints will be confined as much as reasonably practicable. All parking, storage areas, laydown sites, equipment storage, and any other surface-disturbing activities will be confined, to the greatest extent possible, to previously disturbed areas. Additionally, the site footprint/area will be clearly defined and marked to avoid working in areas outside of the approved area. Fences and flagging will be installed by the contractor in a manner that does not impact habitats and other sensitive areas to be avoided and that is clearly visible to personnel on foot and operating heavy equipment. On National Forest System (NFS) lands, the Forest Service will approve the final proposed work area prior to commencement of work.	All
3	Garbage and Microtrash	Work areas will be kept clear of garbage, including microtrash (small pieces of trash or smaller, broken-down pieces of trash). Trash and food will be stored in closed containers and removed daily to reduce attractiveness to opportunistic predators such as coyotes, domestic and feral dogs and cats, opossums, skunks, and raccoons. Littering of trash and food waste will be prohibited. Upon completion of a decommissioning activity, the work site will be inspected to ensure it is free of garbage and microtrash. If garbage or microtrash is detected at the site, it will be removed.	All
4	Construction Timing	Impacts to the community will be minimized, to the extent possible, through the use of seasonally appropriate construction windows.	All

Table 4-1. Measures for Borel F	Project Decommissioning
---------------------------------	-------------------------

Measure No.	Subject	Measure	Applicable Segment
5	Speed Limits	All construction equipment and vehicles will drive no faster than 15 miles per hour anywhere within the Borel FERC Project boundary for reasons of public safety, avoidance of wildlife collisions, and to prevent excess dust. Vehicles will stay on designated roads to the extent reasonably possible. Construction truck trips will be minimized to the extent practicable, particularly in the community and on the grade between Bakersfield and Lake Isabella.	All
General Con	struction Measures		
6	Hazardous Materials	The contractor will be required to provide a Project-specific hazardous materials handling plan prior to start of work. All work-related materials will be properly stored and secured. Materials that are in any type of liquid or powder form will be stored in sealed leak-proof containers. In addition, all parked vehicles/equipment will be kept free of leaks, particularly antifreeze, as this could be fatal if consumed by wildlife. Any proposed use of herbicides on NFS land will require approval of the Forest Service. If used, information on herbicides will be documented and provided to the SQF botanist.	All
7	Hazardous Liquids	The contractor will be required to conduct vehicle refueling and O&M in upland areas, where fuel cannot enter aquatic habitats or areas that have suitable habitat to support Federally and/or State listed species. Equipment and containers will be inspected daily for leaks. Should a leak occur, contaminated soils and surfaces will be cleaned up and disposed as required by the Borel Project's regulatory permits and materials safety data sheets.	All

Table 4-1	. Measures	for Borel	Project	Decommissioning
-----------	------------	-----------	---------	-----------------

Measure No.	Subject	Measure	Applicable Segment
8	Invasive Weeds Prevention	Use certified weed-free straw or rice straw for all construction, erosion control, or restoration needs. Use gravel and sand from local and weed-free sources where possible. Whenever possible, dispose of any spoils on site, graded to match local contours, and use fill collected on site. On NFS lands, SCE will coordinate with the Forest Service on buffers around invasive weed occurrences during construction and conduct a year of post-construction monitoring for invasive weeds within the active work and work-related areas. Additionally, work will generally follow Forest Service Manual 2903 for invasive plant management, as practicable, on NFS lands.	All
9	Construction Plans	SCE or the contractor will develop a suite of plans that the contractor will be required to follow throughout the decommissioning process. These plans are expected to include, but are not limited to, a traffic control plan, a staging and haul route plan, a materials handling plan, a construction safety plan, a specific fire safety plan, a dewatering plan, and a Stormwater Pollution Prevention Plan (SWPPP).	All
10	Use of Local Construction Materials	Construction supplies will be purchased from local businesses to the extent practicable.	All
11	Clean Fill	Imported fill will be minimized to the extent possible. All imported fill will come from clean sites (soils will be chemically tested as needed) and be weed-free.	All
12	Modern Vehicles	On-road heavy duty truck fleet will comply with California Title 13 CCR § 2025, which requires that older vehicles be replaced by modern, emission-controlled trucks.	All

Measure	Subject	Measure	Applicable Segment
No.			
General Con	struction Measures		
13	WEAP	A WEAP will be established and implemented prior to the start of work activities in the field and cover biological and cultural resources. The program will be presented by a qualified biologist and a qualified archaeologist to all construction crew members. If new employees join the crew, they will receive formal, approved training prior to working on site. Upon completion of the orientation, employees will sign a form stating they attended the program and understand all protection measures. A fact sheet containing the presented information will also be prepared and distributed. For biological resources, the WEAP will cover special-status wildlife species, general behavior and ecology of these species, their sensitivity to human activities, their legal protection, penalties for violating Federal laws, reporting requirements, Borel Project mitigation measures, and measures to implement in the event that the species is found during activities. For cultural resources, the WEAP will cover the existence of and potential for cultural and Tribal resources in the Borel Project Vicinity, and contractor roles/responsibilities in the case of an inadvertent discovery during construction	All
Wildlife and	Habitat Measures		
14	ESA Birds and Habitat	No work activities will take place upstream of the Canal Inlet Structure to prevent potential impacts to ESA-listed bird habitat and other sensitive natural communities present in this Borel Project segment.	1
15	Biological Monitor	A biological monitor will be on site during all ground-disturbing and vegetation removal activities associated with the decommissioning in areas of sensitive vegetation communities, ESA-listed species habitat or known special-status species occurrences. On NFS lands, a biological monitor will be present when work occurs near a known NNIP occurrence.	All

Table 4-1. Measures for Borel Project Decommissioning

Measure	Subject	Measure	Applicable Segment	
Wildlife and	Wildlife and Habitat Measures			
16	Pre-construction Surveys	Prior to the start of activities that may impact biological resources in each specific segment of the Borel Project, pre- construction surveys for sensitive habitats and sensitive species, including ESA-listed species and special-status plants on NFS Lands, will be conducted. Surveys will be conducted by qualified biologists and during the appropriate timeframe for detection of target species, within the given period for the activity (e.g., nesting bird surveys will not be performed for activities that will take place completely outside of the nesting bird season). On NFS lands, the designated Forest Service botanist will be consulted for specific types of data and mapping needed and the data collected will be provided to the designated Forest Service botanist. Survey timing will follow guidance described above but be confirmed with the Forest Service on NFS lands.	All	
17	Revegetation	Upon completion of work activities, temporarily disturbed areas will be revegetated with native plant species. A revegetation plan will be developed that addresses revegetating areas where Borel Project features have been removed. The revegetation plan will also detail any proposed non-native invasive species management and monitoring. Monitoring for a year following construction will be a part of the revegetation plan. To the extent possible, restoration of disturbed areas will use locally grown native plants that are weed and pathogen free, and species and seeds purchased will come from a verified weed- free native seed nursery. On NFS lands, any hydroseeding will follow Forest Service prescribed rules.	6, 7, 8, 9, 10, 11	

Table 4-1. Measures for Borel Project Decommissioning

Table 4-1. Measures for Borel Pro	ject Decommissioning
-----------------------------------	----------------------

Measure No.	Subject	Measure	Applicable Segment
18	Reporting Injured, Diseased, or Deceased Wildlife	All decommissioning staff will report any instances of injured, notably diseased, or deceased wildlife observed within the Borel FERC Project boundary to the SCE authorized representative or designee, who will report the information to the appropriate jurisdictional agency(ies).	All
19	Active Bird Nests	To protect native breeding birds, work activities will avoid to the extent possible the general avian breeding season of February 1 through September 15. If decommissioning activities cannot be avoided during this period, a focused survey for active nests within the area proposed for work will be conducted prior to the commencement of Borel Project activities. If no nests are located, work may proceed as planned. If nesting activity is detected, a protective buffer will be established, as determined by a qualified biologist.	All
20	Bat Exclusion	The year prior to the proposed start of the removal of Borel Project facilities with suitable bat habitat, humane exclusion devices will be placed on all Borel Project facilities that will be removed/filled and have known bat occupation, signs of bat occupation, or potential bat habitat. The humane exclusion device will be installed at the appropriate time of year, outside of maternal season (April and late August) and outside of hibernation season (between November and February). Typically, humane exclusion devices should be left in place for a minimum of seven nights and, in some cases, two full weeks to ensure all bats have left the facility. Surveys will be completed by a qualified biologist to ensure humane exclusion devices have worked properly and all bats have left before permanent exclusion devices are installed. A permanent exclusion device must follow a humane exclusion device immediately after the area is bat-free.	1, 2, 4, 6, 7, 9, 10, 11

Table 4-1. Measures for Borel Pro	oject Decommissioning
-----------------------------------	-----------------------

Measure No.	Subject	Measure	Applicable Segment
Wildlife and	Habitat Measures		
21	Bat Surveys	No more than 7 days prior to the removal/fill of Borel Project features where permanent bat exclusion devices have been placed, a qualified biologist will perform a survey of the feature(s) to ensure no bats are present and exclusion devices are still functional. Exclusion devices will only be removed, if necessary, no more than 1 day before decommissioning activities on the feature commence.	1, 2, 4, 6, 7, 9, 10, 11
22	Special-status Species	If special-status species are detected, those individuals will be allowed to move from the area of their own volition. If impacts to special-status species cannot be avoided, the agency(ies) with jurisdiction will be consulted and any necessary permits or approvals will be acquired prior to the commencement of decommissioning activities. Damage or injury to special-status species will be reported immediately to the agency(ies) with jurisdiction.	All
23	Excavations	For any activities requiring an excavation, if excavations are to be left open and unattended for more than 12 hours, they will either be covered, surrounded with exclusion fencing, or an escape ramp will be constructed to the bottom of the pit with less than a 2:1 slope, to provide an escape route to prevent small wildlife species (e.g., lizards, rodents) in the area from getting trapped in the excavation. To the extent feasible, excavations will not be left open at the end of the day and will be covered after confirming absence of trapped individuals. Prior to commencement of work activity each day, staff will check excavations to ensure no animals are trapped. Before backfilling or permanently closing any excavation, it will be checked to ensure no wildlife are present within the excavated area. If wildlife has become trapped, it will be removed prior to closure or backfilling.	All

Table 4-1. Measures for Borel Pre	oject Decommissioning
-----------------------------------	-----------------------

Measure No.	Subject	Measure	Applicable Segment
24	Riparian Vegetation	Riparian vegetation removal and trimming will be limited to the amount necessary to successfully complete activities, including any elderberry shrubs in riparian areas. To prevent unintended or unnecessary removal or trimming of riparian vegetation, orange barrier fencing, or flagging, will be erected to clearly define the habitat to be avoided during work activities. The Forest Service will be consulted on the protection of elderberry shrubs located on NFS lands outside of riparian areas. They will not be afforded extra protections on non-NFS lands.	All
Wildlife and	Habitat Measures		
25	Special-status Plants	Tracy's eriastrum and Kern Canyon clarkia occurrences, or other newly located occurrences of special-status plants on NFS lands, will be flagged and avoided to the extent feasible (to a maximum of 50 feet, if possible, without interfering with necessary work activities). If work is completed during reproductive life stages, a biological monitor should be present periodically to determine if there is damage or removal of the Tracy's eriastrum and Kern Canyon clarkia due to work activities. New occurrences and/or damage or injury to special- status species will be reported immediately to the agency(ies) with jurisdiction. If there is damage, the occurrence will be resurveyed after the completion of work to determine extent.	All

Table 4-1. Measures for Borel Pre	oject Decommissioning
-----------------------------------	-----------------------

Measure No.	Subject	Measure	Applicable Segment
26	Designated Biologist	A designated qualified biologist will review final plans, designate areas that need temporary fencing, and monitor construction activities within and adjacent to areas with aquatic or other sensitive habitats. The qualified biologist will monitor activities within designated areas during critical times, such as initial ground disturbing activities (e.g., ESA fencing installation), and check that all regulatory agency permit requirements, conservation measures, and mitigation measures are properly implemented and followed. The qualified biologist will check construction barriers or exclusion fencing and provide corrective measures to the contractor to keep the barriers or fencing maintained throughout construction.	All
27	Equipment Cleaning	Prior to the first time any vehicles and equipment, including hand tools, enters a work area, a qualified biologist will perform an inspection for NNIP. All visible soil, plant materials, animal remnants, or any other signs of invasive species on vehicles and equipment will be removed prior to entering the Borel Project site. Removal and decontamination requirements of vehicles and equipment will be up to the discretion of a qualified biologist. If contamination is small enough to be managed on site, the qualified biologist may approve the decontamination of the vehicle or equipment at a proper staging area with adequate containment. Any materials removed at a containment site must be bagged and taken off site. If contamination is extensive, the contractor may be required to take the vehicle or equipment to an off-site wash station. Additionally, if a vehicle or piece of equipment must leave the work site for any length of time and has been exposed to a different work site. Vehicles and equipment that perform work in known NNIP occurrences during work activities should be cleaned before leaving the site. The Forest Service will be notified at least five working days prior to equipment being moved on to NFS lands, including information on equipment cleaning.	All

Measure No.	Subject	Measure	Applicable Segment
28	ESA-listed Birds	No work activities will take place within approximately 0.5 mile of any of the mapped potential nesting habitat patches for least Bell's vireo, southwestern willow flycatcher and yellow-billed cuckoo during the avian breeding season (February 1– September 1).	1
29	ESA-listed and CESA-listed Species	If any ESA-listed or CESA-listed species are observed during pre-construction surveys or work activities, SCE will notify USFWS and/or CDFW. All ESA-listed and CSA-listed species will be allowed to leave a work area without harassment.	All
Water Resources and Aquatic Resource Measures			
30	Natural Drainage	Natural landscape drainage patterns will be maintained to the extent practicable.	All
31	Avoidance of Aquatic Habitat	Impacts to delineated aquatic resources, outside of the Borel Canal, will be limited to the amount necessary to successfully complete all work activities. To prevent unintended or unnecessary impacts, orange barrier fencing or flagging will be erected to clearly define the aquatic habitat to be avoided.	All
32	SWPPP	SCE or the contractor will develop a SWPPP in accordance with the State Water Resources Control Board General Construction Permit and local regulations. The SWPPP will include BMPs to reduce or eliminate construction impacts to stormwater runoff. On NFS lands, Forest Service personnel will be present and work alongside the contractor's Qualified SWPPP Developer (QSD)/Qualified SWPPP Practitioner (QSP).	All
33	Lake Isabella Elevation	Work in Lake Isabella will be completed during dry conditions when the lake elevation is at 2,535 feet mean sea level or below.	1, 2, 3, 4

Table 4-1. Measures for Borel Pro	oject Decommissioning
-----------------------------------	-----------------------

Measure No.	Subject	Measure	Applicable Segment
Cultural and	Tribal Resources Measures		
34	Ground Disturbance	Ground disturbance in the vicinity of unevaluated or NRHP- eligible archaeological sites, Traditional Cultural Properties (TCPs), and Traditional Cultural Landscapes (TCLs) will be avoided to the extent possible. All decommissioning-related ground disturbance will be confined to within the Borel FERC Project boundary.	All
35	Avoidance and Exclusion	A qualified archaeologist will review final plans and, in coordination with the appropriate land-managing agency, will designate areas that need temporary exclusion fencing, signage, flagging, barriers, or other avoidance and exclusion measures. Of particular note are the historic-era mining features located both upslope and downslope from the canal near Pioneer Siphon. Prior to any ground disturbance, these features will be relocated and designated for avoidance. The archaeologist will check construction barriers or exclusion fencing and provide corrective measures to the contractor to ensure the barriers or fencing are maintained throughout construction.	All
36	Footings	Footings will be left in place at siphons, flumes, and penstocks to minimize ground disturbance to the extent possible.	All
37	Archaeological Analysis and Consultation	Additional analysis is required to determine effects (if any) related to the decommissioning process. Contingent on the results of the analysis, further measures may be necessary.	All
38	Tribal Consultation	Analysis and consultation with Tribes will continue to accurately characterize the extent of Tribal resources and assess effects of decommissioning activities on previously recorded or newly documented Traditional Cultural Properties (TCPs) and Traditional Cultural Landscapes (TCLs).	All

Table 4-1. Measures for Borel F	Project Decommissioning
---------------------------------	-------------------------

Measure No.	Subject	Measure	Applicable Segment
39	Cultural and Tribal Monitors	On-site cultural monitoring by a qualified archaeologist will be necessary near all unevaluated and NRHP-eligible archaeological sites during decommissioning-related ground disturbance. Tribal monitoring will likely be necessary in any area deemed culturally sensitive by the Tribe(s). Identification of these areas will be borne out of the ongoing consultation effort noted in Measure Nos. 37 and 38.	All
Cultural and	Tribal Resources Measures		
40	Treatment of Historic Properties	Development of an agreement document to resolve adverse effects; agreement document will outline appropriate mitigation to resolve adverse effects. Effects to the Borel Hydroelectric Historic District will include documentation of the district via HABS / HAER documentation and/or equivalent. Mitigation will be developed in consultation with consulting parties	All
41	Inadvertent Discovery	In coordination with consulting parties, develop a Borel Project Inadvertent Discovery and Monitoring Plan that details the protocols to be implemented when necessary, including any specific requirements of the SQF and BLM, in the case of an inadvertent discovery of previously unrecorded archaeological resources. These protocols will include the necessary compliance and reporting requirements for the discovery of human remains on both Federal and non-Federal lands	All

5 References

None